Memorandum to: Supplemental Retirement Board of Trustees

From: Mary Laurie Cece, Assistant General Counsel, North Carolina Department of State Treasurer

Re: Changes in Plan Documents

Date: November 30, 2015

Included in the materials for the December 10, 2015 Board meeting are a restated plan document for the Supplemental Retirement Income Plan of North Carolina (NC 401(k)) and an amended adoption agreement for the North Carolina Public Employee Deferred Compensation Plan (NC 457). Both documents need formal Board approval. Language has been added to these two plan documents that will allow for automatic cash out of participant balances of \$1000 or less. Plan participants that are deemed Bailey vested have been excluded from the automatic cash out, but rather will need to give affirmative approval.

In addition, the NC 401(k) plan document has been amended to allow for automatic enrollment should at some time in the future the legislature approve such a feature. The NC 401(k) will be filing for a final determination letter with the IRS in January and thus it was thought to be prudent to add this language at this time. The NC 457 adoption agreement already has a provision for auto-enroll if the employer so choses. Finally, the NC 401(k) plan document has been updated in some non-substantive ways to enhance consistency throughout the document.

# SUPPLEMENTAL RETIREMENT INCOME PLAN OF NORTH CAROLINA

Effective January 1, 1985 As Amended and Restated Effective December 10, 2015

# SUPPLEMENTAL RETIREMENT INCOME PLAN OF NORTH CAROLINA

#### **PREAMBLE**

The Supplemental Retirement Income Plan A of North Carolina ("Plan A"), the Supplemental Retirement Income Plan B of North Carolina ("Plan B"), and the Supplemental Retirement Income Plan C ("Plan C") were established as of January 1, 1985. Effective as of January 1, 2002, Plan A, Plan B and Plan C were amended to comply with the provisions of the Economic Growth and Tax Relief Reconciliation Act of 2001.

Effective as of the close of business on September 30, 2003, Plan B and Plan C were merged into and succeeded by Plan A. Effective as of October 1, 2003, Plan A was renamed the Supplemental Retirement Income Plan of North Carolina ("Plan") and amended and restated in its entirety. The Plan was amended and restated, effective as of January 1, 2007 to reflect certain design changes and amendments that went into effect after the restatement of the Plan in 2003. The Plan is now amended and restated, effective as of December 10, 2015 to reflect certain design changes and amendments adopted through this December 10, 2015 restatement of the Plan. The Plan is intended to be a qualified profit-sharing plan as that term is defined by Section 401 of the Internal Revenue Code (the "Code") and to be a qualified cash-or-deferred arrangement under Section 401(k) of the Code.

Except as otherwise herein specified, the rights and benefits of any Member who retires or whose employment is terminated before the effective date of the restatement of the Plan or any amendment to the Plan, including such person's eligibility for benefits, shall be determined solely in accordance with the provisions of the Plan in effect and operative at the time of such retirement or termination, unless otherwise required by law, or unless such person is thereafter reemployed and again becomes an employee eligible to participate in the Plan.

# SUPPLEMENTAL RETIREMENT INCOME PLAN OF NORTH CAROLINA

# TABLE OF CONTENTS

|               |                                                                          | <b>Page</b> |
|---------------|--------------------------------------------------------------------------|-------------|
| Article 1. De | efinitions                                                               | 1           |
| Article 2. El | igibility and Membership                                                 | 7           |
| 2.01          | Eligibility                                                              |             |
| 2.02          | Membership                                                               |             |
| 2.03          | Reemployment of Former Employees and Former Members                      |             |
| 2.04          | Transferred Members                                                      |             |
| 2.05          | Termination of Membership                                                |             |
| Article 3. Co | ontributions                                                             | 10          |
| 3.01          | Deferred Cash Contributions                                              |             |
| 3.02          | Employer Special Contributions                                           |             |
| 3.03          | Transfer Contributions                                                   |             |
| 3.04          | Change in Contributions                                                  |             |
| 3.05          | Suspension of Contributions                                              |             |
| 3.06          | Maximum Annual Additions                                                 |             |
| 3.07          | Return of Contributions                                                  | 21          |
| 3.08          | Member Rollover Contributions                                            | 22          |
| 3.09          | Contributions During Period of Military Leave                            | 23          |
| Article 4. In | vestment of Contributions                                                | 26          |
| 4.01          | Investment Funds                                                         | 26          |
| 4.02          | Investment of Members' Accounts                                          | 26          |
| 4.03          | Responsibility for Investments                                           | 27          |
| 4.04          | Change of Election                                                       | 27          |
| 4.05          | Transfers/Reallocation Between Funds                                     | 27          |
| 4.06          | Limitations Imposed by Contract, Prospectus or Other Document of Similar | 20          |
|               | Import                                                                   | 28          |
|               | duation of the Accounts                                                  |             |
| 5.01          | Valuation of the Investment Funds                                        |             |
|               | Discretionary Power of the Primary Administrator                         |             |
| 5.03          | Statement of Accounts                                                    |             |
| 5.04          | Valuation Dates                                                          | 29          |
| Article 6. Ve | ested Portion of Accounts                                                |             |
| 6.01          | Vesting                                                                  | 30          |
| Article 7. W  | ithdrawals While Still Employed                                          | 31          |
| 7.01          | Withdrawal After Age 59½                                                 |             |

| 7.02           | Hardship Withdrawal                                                            |    |
|----------------|--------------------------------------------------------------------------------|----|
| 7.03           | Procedures and Restrictions                                                    | 34 |
| 7.04           | Transfers directly to the Teachers' and State Employees' Retirement System of  |    |
|                | North Carolina, the Consolidated Judicial Retirement System of North Carolina, |    |
|                | the Legislative Retirement System of North Carolina, the North Carolina Local  |    |
|                | Government Employees' Retirement System and the Charlotte Firefighters         |    |
|                | Retirement System.                                                             | 34 |
| Article 8. Loa | nns to Members                                                                 | 37 |
| 8.01           | Amount Available                                                               |    |
| 8.02           | Terms                                                                          | 38 |
| Article 9 Dis  | tribution of Accounts Upon Termination of Employment                           | 41 |
| 9.01           | Eligibility                                                                    |    |
| 9.02           | Forms of Distribution                                                          |    |
| 9.03           | Commencement of Payments                                                       |    |
| 9.04           | Mandatory Distribution of De Minimis Accounts                                  |    |
| 9.05           | Status of Accounts Pending Distribution                                        |    |
| 9.06           | Proof of Death and Right of Beneficiary or Other Person                        |    |
| 9.07           | Distribution Limitation                                                        |    |
| 9.08           | Direct Rollover of Certain Distributions.                                      |    |
| 9.09           | Waiver of Notice Period                                                        |    |
| A-42-1- 10 A   | lucini ducdi cu of Dlou                                                        | 40 |
|                | Iministration of PlanAppointment of Primary Administrator                      |    |
| 10.01          | **                                                                             |    |
| 10.02          | Duties of Primary Administrator                                                |    |
| 10.03<br>10.04 | Individual Accounts                                                            |    |
| 10.04          | Establishment of Rules                                                         |    |
|                |                                                                                |    |
| 10.06<br>10.07 | Prudent Conduct                                                                |    |
|                |                                                                                |    |
|                | anagement of Funds  Trust Agreement                                            |    |
| 11.02          | Exclusive Benefit Rule                                                         |    |
| 11.02          | Investment In Group Trust                                                      |    |
| A 41 1 10 C    | ID                                                                             | 50 |
|                | eneral Provisions                                                              |    |
| 12.01          | Nonalienation                                                                  |    |
| 12.02          | Conditions of Employment Not Affected by Plan                                  |    |
| 12.03          | Facility of Payment                                                            |    |
| 12.04          | Information                                                                    |    |
| 12.05          | Construction                                                                   |    |
| 12.06          | Erroneous Allocation                                                           |    |
| 12.07          | Elections                                                                      |    |
| 12.08          | Plan Expenses                                                                  | 54 |

| Article 13. Ar | nendment, Merger and Termination | . 55 |
|----------------|----------------------------------|------|
|                | Amendment of Plan                |      |
| 13.02          | Merger or Consolidation          | . 55 |
|                | Termination of Plan              |      |
|                |                                  |      |

# Appendix A

# SUPPLEMENTAL RETIREMENT INCOME PLAN OF NORTH CAROLINA

Effective January 1, 1985

As Amended and Restated Effective December 10, 2015

## **ARTICLE 1. DEFINITIONS**

- 1.01 "Accounts" means the Deferred Account, the Employer Account, effective as of June 1, 2006, the Roth Account, the Rollover Account and the Transfer Account.
- 1.02 "Annual Dollar Limit" means the annual dollar limit set forth in Section 401(a)(17)(A) of the Code, as adjusted from time to time for cost of living in accordance with Section 401(a)(17)(B) of the Code.
- 1.03 "Annuity Starting Date" means the first day of the first period for which an amount is paid following a Member's retirement or other termination of employment.
- 1.04 "Beneficiary" means any person or persons named by a Member by written designation filed with the Administrator to receive benefits payable in the event of the Member's death. If no such designation is in effect at the time of death of the Member, or if no person or persons so designated shall survive the Member, the Beneficiary shall be the estate of the Member.
- 1.05 "**Board**" means the Supplemental Retirement Board of Trustees as established under the provisions of N.C.G.S. 135-96.
- 1.06 "Catch-Up Contributions" means, effective January 1, 2002, Tax-Deferred Contributions made to the Plan pursuant to Section 3.01(e), which constitute Catch-Up Contributions under Section 414(v) of the Code. The determination of whether any contribution constitutes a Catch-Up Contribution for a Plan Year shall be determined as of the end of such Plan Year, in accordance with Section 414(v) of the Code and the regulations thereunder.
- 1.07 "Code" means the Internal Revenue Code of 1986, as amended from time to time.

- 1.08 "Compensation" means all salaries and wages prior to any reduction pursuant to Sections 125, 132(f), 401(k), 403(b), 414(h)(2), and 457 of the Code, not including any terminal payments for unused sick leave derived from public funds, which are earned by a Member while an Employee of an Employer. Compensation shall not include any payment, as determined by the Board of Trustees, for the reimbursement of expenses or payments for housing or any other allowances whether or not classified as salary and wages. However, Compensation for a Plan Year shall not exceed the Annual Dollar Limit. Notwithstanding the foregoing, effective as of January 1, 2004, with respect to a Member who is a member of the General Assembly contributing to the Legislative Retirement System, "Compensation" means salary and expense allowance paid to a Member for service as a legislator in the North Carolina General Assembly, exclusive of travel and per diem. Notwithstanding the foregoing, effective January 1, 2009, Compensation shall include differential wage payments within the meaning of Section 3401(h) of the Code.
- 1.09 "**Deferred Account**" means the account into which shall be credited the Tax-Deferred Contributions made on a Member's behalf and earnings on those contributions.
- 1.10 **"Deferred Cash Contributions"** means the Tax-Deferred Contributions and Roth Contributions contributed pursuant to Section 3.01.
- 1.11 "Earnings" means the amount of earnings to be returned with any excess deferrals under Section 3.01 determined in accordance with regulations prescribed by the Secretary of Treasury under the provisions of Section 402(g) of the Code.
- 1.12 "Effective Date" means January 1, 1985.

## 1.13 "**Employee**" means

- (a) with respect to Part "A", a person who is a contributing member of the Teachers' and State Employees' Retirement System of North Carolina (who is not a state-employed law enforcement officer), the Consolidated Judicial Retirement System or the Legislative Retirement System or a person who is a contributing participant in the Optional Retirement Program;
- (b) with respect to Part "B", a person who is a contributing member of the Local Governmental Employees' Retirement System, a person who is a law enforcement officer

employed by any political subdivision of the State of North Carolina or a person who is an employee of a political subdivision of the State of North Carolina and is a participating member of a retirement or pension plan qualified under Section 401(a) of the Code which is sponsored by a political subdivision of the State of North Carolina; and

- (c) with respect to Part "C", a person who is employed by the State of North Carolina as a law enforcement officer and who is (i) a contributing member of the Teachers' and State Employees' Retirement System of North Carolina or (ii) eligible for allocations under another pension plan qualified under Section 401(a) of the Code, which is sponsored by a political subdivision.
- 1.14 "**Employer**" means, with respect to its employees, the State of North Carolina, any employer as that term is defined in N.C.G.S. 128-21(11) or N.C.G.S. 135-1(11) and any other political subdivision of the state or other entity under the control of the state or one of its political subdivisions as approved by Board of Trustees.
- 1.15 "**Employer Account**" means the account into which shall be credited the Employer contributions made on a Member's behalf pursuant to Section 3.02 and earnings on those contributions.
- 1.16 "Enrollment Date" means any business day of a calendar month.
- 1.17 "**Fund**" or "**Investment Fund**" means the separate funds in which contributions to the Plan are invested in accordance with Article 4.
- 1.18 "Highly Compensated Employee" means for a Plan Year commencing on or after January 1, 1997, any employee of the Employer (whether or not eligible for membership in the Plan) who for the preceding Plan Year received Statutory Compensation in excess of \$80,000. The \$80,000 dollar amount in the preceding sentence shall be adjusted from time to time for cost of living in accordance with Section 414(q) of the Code.

The provisions of this Section shall be further subject to such additional requirements as shall be described in Section 414(q) of the Code and its applicable regulations, which shall override any aspects of this Section inconsistent therewith.

- 1.19 "Member" means any person included in the membership of the Plan as provided in Article 2.
- 1.20 "Nonhighly Compensated Employee" means for any Plan Year an employee of the Employer who is not a Highly Compensated Employee for that Plan Year.
- 1.21 "**Notice**" means the indication by the Employee of his wishes through the means written, electronic, or telephone, provided for the particular purpose by the Primary Administrator.
- 1.22 "Plan" means the Supplemental Retirement Income Plan of North Carolina which is comprised of
  - (a) Part "A" covering Employees who meet the requirements provided in item 1 of Appendix A:
  - (b) Part "B" covering Employees who meet the requirements provided in item 2 of Appendix A; and
  - (c) Part "C" covering Employees who meet the requirements provided in item 3 of Appendix A,

as set forth in this document or as amended from time to time.

- 1.23 "Plan Year" means the 12-month period beginning on any January 1.
- 1.24 "**Primary Administrator**" means the Retirement Systems Division of the North Carolina Department of the State Treasurer.
- 1.25 "Retirement System" means the Teachers' and State Employees' Retirement System of North Carolina, the Consolidated Judicial Retirement System, the Legislative Retirement System and the Local Governmental Employees' Retirement System.
- 1.26 "Rollover Account" means the account credited with Rollover Contributions made by the Member and earnings on those contributions. Beginning January 1, 2002, before-tax amounts rolled over from an eligible deferred compensation plan under Section 457(b) of the Code which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state

or political subdivision of a state shall be accounted for separately within the Member's Rollover Account. Beginning January 1, 2002, after-tax amounts rolled over from a plan qualified under Section 401(a) of the Code shall be accounted for separately within the Member's Rollover Account. Effective on and after June 1, 2006, amounts attributable to Roth Contributions directly rolled over from a plan qualified under Section 401(a) of the Code shall be credited to a Member's Roth Account (or subaccount thereof).

- 1.27 "Rollover Contributions" means amounts contributed pursuant to Section 3.08.
- 1.28 **"Roth Account"** means the account credited with Roth Contributions, and earnings on those contributions. Unless otherwise indicated, any reference in the Plan to the Roth Account shall be deemed to refer as well to the In-Plan Roth Conversion Subaccount described in Sections 7.05 and 9.10.
- 1.29 **"Roth Contributions"** means the amount of Deferred Cash Contributions contributed under Section 3.01 that the Member elected to include in gross income at the time deferred pursuant to Section 3.01(c).
- 1.30 "**Special Contributions**" means all amounts contributed pursuant to Section 3.02 of the Plan.
- 1.31 "**Tax-Deferred Contributions**" means all amounts contributed pursuant to Section 3.01 of the Plan.
- 1.32 "**Transfer Account**" means the account into which shall be credited the Transfer Contributions made by a Member and earnings on those contributions.
- 1.33 "**Transfer Contributions**" means the contributions which were transferred to the Plan by the Member from another qualified plan pursuant to Section 3.03.
- 1.34 "**Trust**" or "**Trust Fund**" means the fund established by the Board as part of the Plan into which benefits are to be paid in accordance with the terms of the Plan.
- 1.35 "**Trustees**" means the trustees holding the funds of the Plan as provided in Section 11.01.

1.36 "Valuation Date" means the date or dates in each calendar month on which any valuation is made, as determined by the Primary Administrator in accordance with procedures established pursuant to Section 5.04

#### ARTICLE 2. ELIGIBILITY AND MEMBERSHIP

## 2.01 Eligibility

Each employee shall be eligible to become a Member on any Enrollment Date coincident with or next following the date on or after the date he becomes an Employee of an Employer participating in the Plan (an "Eligible Employee").

Notwithstanding the foregoing, the eligibility for the Plan of Employees who otherwise would be eligible to become Members can be delayed by the Primary Administrator until the Primary Administrator is satisfied that the payroll systems applicable to those Employees have been significantly modified to permit such Employees to contribute to the Plan.

# 2.02 Membership

- (a) Each person who was a Member under the Plan on December 9, 2015 shall continue as a Member on December 10, 2015, subject to the provisions of Section 2.05.
- (b) Each other person who is an Eligible Employee shall become a Member on the first Enrollment Date which is at least 60 days, or such shorter period as the Employer determines, after the date he files with the Employer a Notice in accordance with procedures prescribed by the Primary Administrator on which he:
- (i) makes the election described in Section 3.01;
- (ii) authorizes the Employer to reduce his Compensation;
- (iii) makes an investment election; and
- (iv) names a Beneficiary,

or, if earlier, the date on which a Special Contribution under Section 3.02 is made on his behalf.

## 2.03 Reemployment of Former Employees and Former Members

Any person reemployed by the Employer as an Eligible Employee, who was previously a Member or who was previously eligible to become a Member shall be immediately eligible to become a Member of the Plan and shall be eligible to make Tax-Deferred Contributions upon the filing of a Notice in accordance with Section 2.02. Any person reemployed by the Employer as an Eligible Employee, who was not previously eligible to become a Member, shall become a

Member upon completing the eligibility requirements described in Section 2.01 and the membership requirements in Section 2.02.

#### 2.04 Transferred Members

A Member who remains in the employ of the Employer but ceases to be an Eligible Employee shall continue to be a Member of the Plan but shall not be eligible to make contributions or receive allocations of Tax-Deferred Contributions or Employer contributions while his employment status is other than as an Employee.

## 2.05 Termination of Membership

A Member's membership shall terminate on the date he is no longer employed by an Employer unless the Member is entitled to benefits under the Plan, in which event his membership shall terminate when those benefits are distributed to him.

#### 2.06 Automatic Enrollment

- (a) General Rule. If automatic enrollment is authorized by the General Assembly, and except as provided in Section 2.06(c) below, if an Eligible Employee has not affirmatively elected either to make or not to make Deferred Cash Contributions under Section 3.01, such Eligible Employee is deemed to have elected to become a Member and have an enrollment application filed on his behalf. The Member will be deemed to have elected to defer a set percentage of his Compensation to the Plan, in the percentage established by the General Assembly and as set forth in procedures of the Primary Administrator. An Eligible Employee will have a reasonable period of time, as established by the Primary Administrator, after receipt of any notice required by the Code, Treasury Regulations, and other applicable guidance, to make an affirmative election regarding Deferred Cash Contributions before the deemed election to make such contributions shall become effective.
- (b) **Default Investment**. When an Investment Fund has not been affirmatively selected by the Member, contributions made under the deemed election of Section 2.06(a) shall be invested in a default Investment Fund determined by the Primary Administrator and the Board.

(c) **Employee Election**. This Section 2.06 shall not apply, or shall cease to apply, to the extent an Eligible Employee files an enrollment application or elects to have no Deferred Cash Contributions contributed to the Plan.

#### **ARTICLE 3. CONTRIBUTIONS**

#### 3.01 Deferred Cash Contributions

- (a) A Member may elect on his application filed under Section 2.02 to reduce by a specified percentage or whole dollar amount his Compensation payable while a Member and have that amount contributed to the Plan by the Employer. Such total amount contributed shall not exceed 80% of such Member's Compensation (or said lesser percentage as the Primary Administrator may determine). That amount shall be contributed to the Plan by the Employer as Deferred Cash Contributions in a manner to be determined by the Primary Administrator. Any Deferred Cash Contributions elected under this Section 3.01 shall be allocated to the Member within the Plan Year for which they are contributed and shall be paid to the Trustee as soon as practicable. Deferred Cash Contributions shall be further limited as provided below and in Section 3.06.
- (b) Effective for taxable years beginning after December 31, 2001, a Member is eligible to make Deferred Cash Contribution under Section 3.01(a) of the Plan and who has attained or will attain age 50 by the last day of the taxable year may elect, in accordance with procedures prescribed by the Primary Administrator, to make Catch-Up Contributions for that taxable year in accordance with and subject to the limitations of Section 414(v) of the Code. Such Catch-Up Contributions shall be subject to the following special rules:
  - (i) A Member's Catch-Up Contributions shall not be taken into account for purposes of applying the limitations under Sections 402(g) and 415 of the Code.
  - (ii) The determination of whether a Deferred Cash Contribution under this Section constitutes a Catch-Up Contribution for any taxable year shall be determined as of the end of such taxable year, in accordance with Section 414(v) of the Code. Deferred Cash Contributions that are intended to be Catch-Up Contributions for a taxable year but which do not qualify as Catch-Up Contributions as of the end of the taxable year shall be treated for all purposes under the Plan as Deferred Cash Contributions made under Section 3.01(a).
  - (iii) In the event that the sum of a Member's Catch-Up Contributions and similar contributions to any other qualified defined contribution plan maintained by the Employer exceeds the dollar limit on Catch-Up Contributions under

Section 414(v) of the Code for any taxable year as in effect for such taxable year, the Member shall be deemed to have elected a return of the Catch-Up Contributions in excess of the limit under Section 414(v) of the Code and such amount shall be treated in the same manner as "excess deferrals" under Section 3.01(e).

- (iv) If a Member makes Catch-Up Contributions under a qualified defined contribution plan maintained by an employer other than the Employer and/or Code Section 403(b) plan for any taxable year and those contributions when added to his Catch-Up Contributions exceed the dollar limit on Catch-Up Contributions under Section 414(v) of the Code for that taxable year, the Member may allocate all or a portion of such "excess Catch-Up Contributions" to this Plan. In the event such Member notifies the Administration of the "excess Catch-Up Contributions" in the same manner as is required for allocated "excess deferrals" under Section 3.01(d), such "excess Catch-Up Contributions" shall be distributed in the same manner as "excess deferrals" under Section 3.01(f).
- (v) The Employer shall not take a Member's Catch-Up Contributions into account for purposes of determining the amount of Special Contributions under Section 3.02.
- (vi) A Member's Catch-Up Contributions shall be subject to the same withdrawal and distribution restrictions as Deferred Cash Contributions made under Section 3.01(a).
- (c) Unless a Member makes an election under the provisions of this paragraph (c), Deferred Cash Contributions and Catch-Up Contributions made by a Member under paragraphs (a) and (b) above shall be deemed to be Pre-Tax Contributions. Effective as of June 1, 2006, in lieu of making Deferred Cash Contributions and Catch-Up Contributions on a pre-tax basis pursuant to the provisions of paragraphs (a) and (b) above, a Member may elect, in accordance with procedures prescribed by the Primary Administrator, to have some or all of the Deferred Cash Contributions and/or Catch-Up Contributions that otherwise would be contributed to the Plan on a pre-tax basis designated as Roth Contributions or Roth Catch-Up Contributions, as applicable, and included in his gross income at the time of deferral. Such election, once made, may only be revoked with respect to Deferred Cash Contributions to be contributed after the effective date of the revocation election. Notwithstanding the foregoing, Members' opportunity to make Roth Contributions can be delayed by the Primary Administrator until the Primary Administrator is satisfied that

the payroll systems applicable to those Employees have been significantly modified to permit such Employees to individual choose whether their contributions to the Plan will be made as Tax-Deferred Contributions or Roth Contributions.

- (d) In no event shall the Member's Deferred Cash Contributions and similar contributions made on his behalf by the Employer to all plans, contracts or arrangements subject to the provisions of Section 401(a)(30) of the Code in any calendar year exceed the applicable dollar limitation contained in Section 402(g) of the Code in effect for such calendar year, except as permitted under Section 414(v) of the Code. If a Member's Deferred Cash Contributions in a calendar year reach that dollar limitation, his election of Deferred Cash Contributions for the remainder of the calendar year will be cancelled. As of the first pay period of the calendar year following such cancellation, the Member's election of Deferred Cash Contributions shall again become effective in accordance with his previous election, unless the Member elects otherwise in accordance with Section 3.04.
- (e) In the event that the sum of the Deferred Cash Contributions and similar contributions to any other qualified defined contribution plan maintained by the Employer exceeds the dollar limitation on elective deferrals under Section 402(g) of the Code for any calendar year as in effect for such calendar year, the Member shall be deemed to have elected a return of Tax-Deferred Contributions in excess of such limit ("excess deferrals") from this Plan. A Member who has made both Tax-Deferred Contributions and Roth Contributions for the applicable calendar year may designate, under procedures prescribed to the Primary Administrator, whether such excess deferrals shall be attributed first to his Tax-Deferred Contributions or his Roth Contributions. In the event a Member fails to make such an election, the excess deferrals shall be attributed first to his Tax-Deferred Contribution. In the event a Member has made both Tax-Deferred Contributions and Roth Contributions for the applicable calendar year, the excess deferrals shall be first attributed to the Member's Tax-Deferred Contributions. The excess deferrals, together with Earnings, shall be returned to the Member no later than the April 15 following the end of the calendar year in which the excess deferrals were made. In the event any Tax-Deferred or Roth Contributions returned under this paragraph (c) were matched by Special Contributions under Section 3.02,

those Special Contributions, together with Earnings, shall be forfeited and used to reduce Employer contributions.

(f) If a Member makes tax-deferred contributions under (i) another qualified defined contribution plan maintained by an employer other than the Employer and/or (ii) any Code Section 403(b) plan for any calendar year and those contributions when added to his Deferred Cash Contribution exceed the dollar limitation on elective deferrals under Section 402(g) of the Code for that calendar year, the Member may allocate all or a portion of such excess deferrals to this Plan. A Member who has made both Tax-Deferred Contributions and Roth Contributions for the applicable calendar year may designate, under procedures prescribed to the Primary Administrator, whether such excess deferrals shall be attributed first to his Tax-Deferred Contributions or his Roth Contributions. In the event a Member fails to make such an election, the excess deferrals shall be attributed first to his Tax-Deferred Contribution. In the event a Member has made both Tax-Deferred Contributions and Roth Contributions for the applicable calendar year, the excess deferrals shall be first attributed to the Member's Tax-Deferred Contributions. In that event, the excess deferrals, with Earnings thereon, shall be returned to the Member no later than the April 15 following the end of the calendar year in which the excess deferrals were made. However, the Plan shall not be required to return excess deferrals unless the Member notifies the Primary Administrator, in writing, by March 1 of that following calendar year of the amount of the excess deferrals allocated to this Plan. In the event any Deferred Cash Contribution returned under this paragraph (d) were matched by Special Contributions, those Special Contributions, together with Earnings, shall be forfeited and used to reduce Employer Contributions.

#### 3.02 Employer Special Contributions

(a) An Employer may make Special Contributions to the Plan each Plan Year on behalf of each Employee who (i) is a Member or eligible to become a Member pursuant to the provisions of Section 2.01, (ii) was actively employed by such Employer during such Plan Year, and (iii) fulfilled such period of eligibility service to receive such contribution as the Employer shall determine (provided that such period shall not in any case exceed one year), in an amount to be determined by the General Assembly or such other governing body which has such authority. These contributions shall be promptly paid to the Trustee by the Employer. Each Employer's Special Contributions under this

paragraph (a), if any, for each Plan Year shall be allocated among the eligible Employees of that Employer on a per capita basis or based on the ratio of its eligible Employee's Compensation during the Plan Year to the total Compensation during that Plan Year for all the eligible Employees of that Employer, as the General Assembly or such other governing body shall so determine. Notwithstanding the foregoing, the General Assembly or such other governing body which has such authority may provide that, effective on or after January 1, 2007, the amount of Special Contributions made by each Employer pursuant to this paragraph (a) may be determined separately for separate groups of such Employer's eligible Employees (as defined pursuant to the provisions of the preceding sentence); provided, however, that (i) each separate group shall consist of a 'reasonable class' of the Employer's eligible Employees as determined by the Primary Administrator and (ii) each Employer shall not create more than three separate groups of eligible Employees during any Plan Year for purposes of making Special Contributions under this paragraph (a) for such Plan Year. Each separate group of an Employer's eligible Employees shall be determined solely on the basis of years of service completed by such eligible Employees while employed by that Employer. Years of service completed by an eligible Employee shall be determined as of a specific date or as of an applicable Plan Year in a manner uniformly consistent for all eligible Employees of the Employer. Notwithstanding the foregoing, a separate group of an Employer's eligible Employees shall not be deemed a 'reasonable class' of the Employer's eligible Employees if such separate group consists of less than 20 percent of all eligible Employees of the Employer, unless such separate group consists of all eligible Employees of the Employer employed as of a specific date occurring on or after the later of January 1, 2007 or the date that separate group is first established for that Employer by the General Assembly or such other governing body. Each Employer's Special Contributions under this paragraph (a), if any, for each Plan Year shall be allocated among the group of eligible Employees of that Employer on a uniform basis, either per capita or according to the ratio of such eligible Employee's Compensation during the Plan Year to the total Compensation during the Plan Year for such group of eligible Employees of that Employer, as the General Assembly or such other governing body shall so determine, and the allocation rates applicable to the separate groups of each Employer shall be based on a 'smoothly increasing schedule' or a 'smoothly

decreasing schedule.' A 'smoothly increasing schedule' is a schedule under which the allocation rate for each separate group of the Employer's eligible Employees does not exceed the allocation rate for the immediately preceding group (that is, the group with the next lower number of years of service) by more than 100 percent for the same Plan Year. A 'smoothly decreasing schedule' is a schedule under which the allocation rate for each separate group of the Employer's eligible Employees for the same Plan Year is not less than one half of the allocation rate for the immediately preceding group (that is, the group with the next lower number of years of service). These contributions shall be promptly paid to the Trustee by the Employer Notwithstanding the foregoing, with the permission of the Employer and under rules uniformly applicable to all employees similarly situated, an Employee may elect to have the Special Contributions allocated on his behalf under this paragraph (a) not contributed to this Plan, but rather contributed on his behalf to another eligible retirement plan (as defined in Sections 402(c)(8)(B)(iii)-(vi) of the Code) sponsored by the Employer.

(b) The Employer may make Special Contributions to the Plan on behalf of each Member who (i) has Deferred Cash Contributions made on his behalf during the Plan Year pursuant to an election under Section 3.01(a), and (ii) has fulfilled such period of eligibility service to receive such contribution as the Employer shall determine (provided that such period shall not in any case exceed one year) in an amount to be determined by the General Assembly or such other governing body which has such authority. These contributions shall be promptly paid to the Trustee by the Employer. The Special Contributions under this paragraph (b), if any, for each Plan Year shall be allocated to the Employer's eligible Members who have Tax-Deferred Contributions made on their behalf during the Plan Year, on a uniform basis, in proportion to their Tax-Deferred Contributions made during the Plan Year for which the Special Contribution is being made. Each Special Contribution under this paragraph (b) shall consist of contributions by each Employer in proportion to the aggregate Tax-Deferred Contributions made pursuant to Section 3.01(a), if any, of the eligible Members in its service. Special Contributions under this paragraph (b) are made expressly conditioned on the Plan satisfying the provisions of Section 3.01(b). If any portion of the Tax Deferred Contributions to which the Special Contribution relates is returned to the Member under 3.01, the corresponding Special Contributions shall be forfeited. Notwithstanding the foregoing, the General Assembly or such other governing body

which has such authority may provide that, effective on or after January 1, 2007, the amount of Special Contributions made by each Employer pursuant to this paragraph (b) may be determined separately for separate groups of such Employer's eligible Members (as defined pursuant to the preceding provisions of this paragraph (b)); provided, however, that (i) each separate group shall consist of a 'reasonable class' of each Employer's eligible Members as determined by the Primary Administrator, and (ii) each Employer shall not create more than three separate groups of eligible Members during any Plan Year for purposes of making Special Contributions under this paragraph (b) for such Plan Year. Each separate group of an Employer's eligible Members shall be determined solely on the basis of years of service completed by such eligible Members while employed by that Employer. Years of service completed by an eligible Member shall be determined as of a specific date or as of an applicable Plan Year in a manner uniformly consistent for all eligible Members of the Employer. Notwithstanding the foregoing, a separate group of the Employer's eligible Members shall not be deemed a 'reasonable class' of the Employer's eligible Members if such separate group consists of less than 20 percent of all eligible Members of the Employer, unless such separate group consists of all eligible Members of the Employer employed as of a specific date occurring on or after the later of January 1, 2007 or the date that separate group is first established for that Employer by the General Assembly or such other governing body. Each Employer's Special Contributions under this paragraph (b), if any, for each Plan Year shall be allocated to the Employer's group of eligible Members who have Tax-Deferred Contributions made on their behalf during the Plan Year on a uniform basis, as the General Assembly or such other governing body shall so determine, in proportion to their Tax-Deferred Contributions made during the Plan Year for which the Special Contribution is being made, and the allocation rates applicable to the separate groups of each Employer shall be based on either a 'smoothly increasing schedule' or 'smoothly decreasing schedule' as defined below. Each Special Contribution under this paragraph (b) shall consist of contributions by each Employer in proportion to the aggregate Deferred Cash Contributions made pursuant to Section 3.01(a), if any, of the group of eligible Members in its service. A 'smoothly increasing schedule' is a schedule under which the allocation rate for each separate group of the Employer's eligible Members does not exceed the allocation rate for the immediately preceding group (that is, the group with the next lower number of years of service) by more than 100 percent for the same Plan Year. A 'smoothly decreasing schedule' is a schedule under which the

allocation rate for each separate group of the Employer's eligible Members for the same Plan Year is not less than one half of the allocation rate for the immediately preceding group (that is, the group with the next lower number of years of service). These contributions shall be promptly paid to the Trustee by the Employer. In the event those Special Contributions subject to forfeitures have been distributed to the Member, the Employer shall make reasonable efforts to recover the contributions from the Member.

- (c) With respect to Employees who are law enforcement officers and are Members or eligible to become Members of Part "B" or Part "C", the Employer shall contribute to the Plan on behalf of each such Member or eligible Member an amount equal to such percentage, as determined by the General Assembly, of Compensation received while a law enforcement officer during such Plan Year. These contributions shall be made at least monthly and shall be promptly paid to the Trustee by the Employer. In addition, any contributions, plus earnings thereon, made by an Employer under this paragraph on account of a mistake of fact and remaining in the Plan after application of the provisions of Section 3.07 shall be allocated as of a date determined by the Primary Administrator on behalf of each Employee who is a law enforcement officer and is a Member or eligible Member on such date on a per capita basis, in equal shares.
- (d) With respect to Employees who are law enforcement officers (not including Sheriffs) and are Members or eligible to become Members of Part "B" or Part "C", there shall be contributed to such Plan on behalf of each such Member or eligible Member on a per capita basis, in equal shares, an amount equal to a division, as determined by the General Assembly, of receipts collected under North Carolina General Statute 7A-304 on account of the assessed cost of court while a Member during such Plan Year. In addition, any contributions, plus earnings thereon, made by an Employer under this paragraph on account of a mistake of fact and remaining in the Plan after application of the provisions of Section 3.07 shall be allocated as of a date determined by the Primary Administrator on behalf of each Employee who is a law enforcement officer (not including Sheriffs) and is a Member or eligible Member on such date on a per capita basis, in equal shares.
- (e) Notwithstanding the foregoing, effective January 1, 2010, an Employer may make Special Contributions to the Plan for a Member who dies or becomes disabled while in

qualified military service pursuant to and in a manner consistent with Code section 414(u)(9) subject to the calculation rules set forth in this Section 3.02.

#### 3.03 Transfer Contributions

- (a) The Primary Administrator under such conditions as the Board may require but without regard to any limitations on contributions set forth in Section 3.06 shall direct the Trustee to accept an amount transferred from the Special Annuity Savings Fund of the Law Enforcement Officers' Retirement System pursuant to N.C.G.S. § 143-166.30 and 143-166.70, or any other amount as determined by the Board. Such Transfer Contributions shall be paid to the Trustee as soon as practicable and shall be held in the Member's Transfer Account under this Plan and invested in the short term investment fund or such other fund as the Primary Administrator determines until such time as a Member makes another investment election as set forth in Section 4.02.
- (b) The Plan may permit, under such conditions as the Primary Administrator may require, an amount to be transferred on behalf of a Member who is not employed by an Employer on the date of the transfer from the Plan to another plan which is qualified under Section 401(a) of the Code.

## 3.04 Change in Contributions

Subject to the provisions of Section 3.01, a Member may change the amount of his authorized payroll reduction by giving such advance Notice as the Primary Administrator shall require. The changed amount shall become effective as soon as administratively practicable following receipt of such Notice.

## 3.05 Suspension of Contributions

- (a) A Member may revoke his election under Section 3.01 by giving such advance Notice the Primary Administrator shall require. The revocation shall become effective as soon as administratively practicable following receipt of such Notice.
- (b) A Member who has revoked his election under Section 3.01 may apply to the Primary Administrator to have his Compensation reduction resumed in accordance with Section 3.01 as soon as administratively practicable following the date the Notice is received by the Primary Administrator.

#### 3.06 Maximum Annual Additions

- (a) The annual addition to a Member's Accounts for any Plan Year, which shall be considered the "limitation year" for purposes of Section 415 of the Code, when added to the Member's annual addition for that Plan Year under any other qualified defined contribution plan of the Employer, shall not exceed an amount which is equal to the lesser of (i) percentage limitation (as set forth in Section 415(c)(1)(B) of the Code) of his Statutory Compensation for that Plan Year or (ii) the dollar limitation in effect under Section 415(c)(1)(A) of the Code for that Plan Year, as adjusted for increases in the cost of living under Section 415(d) of the Code.
- (b) For purposes of this Section, the "annual addition" to a Member's Accounts under this Plan or any other qualified defined contribution plan maintained by the Employer shall be the sum of:
  - (i) the total contributions, including Deferred Cash Contributions, made on the Member's behalf by the Employer,
  - (ii) all member contributions, exclusive of any Transfer Contributions or Rollover Contributions, and
  - (iii) forfeitures, if applicable, and
- (iv) solely for purposes of clause (i) of paragraph (a) above, amounts described in Sections 415(1)(1) and 419A(d)(2) allocated to the Member that have been allocated to the Member's Accounts under this Plan or his accounts under any other such qualified defined contribution plan maintained by an Employer. For purposes of this paragraph (b), any Special Contributions distributed or forfeited under the provisions of Section 3.01, shall be included in the annual addition for the year allocated.

Annual additions shall not include Rollover Contributions, loan repayments made under Article 8, excess deferrals timely distributed from the Plan under Section 3.01(e) or (f), and Catch-Up Contributions.

(c) For purposes of this Section, the term "Statutory Compensation" with respect to any Member shall mean the wages, salaries, and other amounts paid in respect of such Member by the Employer for personal services actually rendered, determined after any reduction of Compensation pursuant to Section 414(h)(2) of the Code, and including (but

limited to) overtime, shift differentials, longevity payments, and amounts contributed by the Employer pursuant to a salary reduction agreement which are not includible in the gross income of the employee under Section 125, 132(f), 402(g)(3), 414(v), or 457(b) of the Code, but excluding deferred compensation and other distributions which receive special tax benefits under the Code. Effective for Plan Years beginning on or after July 1, 2007, Statutory Compensation (i) shall include amounts required to be recognized under the provisions of U.S. Treasury Department regulation 1.415(c)-2(e) and (ii) shall not exceed the limitation on compensation under Section 401(a)(17) of the Code. Effective January 1, 2009, Statutory Compensation shall include differential wage payments within the meaning of Section 3401(h) of the Code.

- (d) If the annual addition to a Member's Accounts for any Plan Year beginning prior to July 1, 2007, prior to the application of the limitation set forth in paragraph (a) above, exceeds that limitation due to a reasonable error in estimating a Member's annual compensation or in determining the amount of Deferred Cash Contributions that may be made with respect to a Member under Section 415 of the Code, or as a result of the allocation of forfeitures, the amount of contributions credited to the Member's Accounts in that Plan Year shall be adjusted to the extent necessary to satisfy that limitation in accordance with the following order of priority:
  - (i) The Member's unmatched Tax-Deferred Contributions under Section 3.01(a) shall be reduced to the extent necessary, with the reduced amount and any earnings thereon returned to the Member.
  - (ii) The Member's unmatched Roth Contributions under Section 3.01(c) shall be reduced to the extent necessary, with the reduced amount and any earnings thereon returned to the Member
  - (iii) The Member's matched Tax-Deferred Contributions and corresponding Special Contributions under Section 3.02(b) shall be reduced to the extent necessary. The amount of the reduction attributable to the Member's matched Tax-Deferred Contributions shall be returned to the Member together with any earnings on those contributions to be returned, and the amount attributable to the Special Contributions shall be forfeited and used to reduce subsequent contributions payable by the Employer.
  - (iv) The Member's matched Roth Contributions and corresponding Special Contributions shall be reduced to the extent necessary. The amount of the

reduction attributable to the Member's matched Roth Contributions shall be returned to the Member together with any earnings on those contributions to be returned, and the amount attributable to the Special Contributions shall be forfeited and used to reduce subsequent contributions payable by the Employer

(v) The Special Contributions, if any, allocated to the Member under Section 3.02(a) or (c) shall be reduced and the amount of the reduction shall be reallocated to the other Members based on the ratio of each other Member's Compensation during the Plan Year to the total Compensation during that Plan Year for all other Members.

Notwithstanding the foregoing, effective for Plan Years beginning on and after July 1, 2007, to the extent that the annual additions to a Member's Accounts exceed the limitation set forth in paragraph (a), corrections shall be made in a manner consistent with the provisions of the Employee Plans Compliance Resolution System as set forth in Revenue Procedure 2013-12 or any subsequent guidance.

(e) Any Deferred Cash Contributions returned to a Member under this paragraph shall be disregarded in applying the dollar limitation on Deferred Cash Contributions under Section 3.01(d).

# 3.07 Return of Contributions

- (a) The Employer may recover without interest the amount of its contributions to the Plan made on account of a mistake of fact, reduced by any investment loss attributable to those contributions, if recovery is made within one year after the date of those contributions.
- (b) In the event that Deferred Cash Contributions made under Section 3.01 are returned to the Employer in accordance with the provisions of this Section 3.07, the elections to reduce Compensation which were made by Members on whose behalf those contributions were made shall be void retroactively to the beginning of the period for which those contributions were made. The Deferred Cash Contributions so returned shall be distributed in cash to those Members for whom those contributions were made.

#### 3.08 Member Rollover Contributions

With the permission of the Primary Administrator and without regard to any limitations on contributions set forth in this Article 3, the Plan may receive from a Member a Rollover Contribution, in cash, consisting of any amount previously received (or deemed to be received) by him from an eligible retirement plan. Such Rollover Contribution shall be subject to the following:

- (a) For purposes of this Section, "eligible retirement plan" means, effective after December 31, 2001:
  - (i) a qualified plan described in Section 401(a) of the Code;
  - (ii) an annuity plan described in Section 403(a) of the Code;
  - (iii) an individual retirement account or individual retirement annuity of the Employee described in Section 408(a) or 408(b) of the Code which contains only amounts that were originally distributed from a qualified plan described in Section 401(a) or 403(a) of the Code (i.e., a "conduit IRA");
  - (iv) an annuity contract described in Section 403(b) of the Code; and
  - (v) an eligible plan under Section 457(b) of the Code which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state.

For purposes of this Section, the term "eligible retirement plan" shall also include, effective on and after October 1, 2011, the following:

- (i) a simplified employee pension described in Section 408(k) of the Code; and
- (ii) a simple retirement account described in Section 408(p) of the Code; provided the amounts received from such simplified employee pension plan or simple retirement account satisfy the provisions of Section 408(d) of the Code regarding rollover contributions.
- (b) Such Rollover Contribution may be received in either of the following ways:
  - (i) The Plan may accept such amount as a direct rollover of an eligible rollover distribution from an eligible retirement plan, including after-tax amounts provided such after-tax amounts are received directly from a plan that is qualified under Section 401(a) of the Code (or effective as of January 1, 2007, an annuity contract defined in Section 403(b) of the Code).
  - (ii) The Plan may accept such amount directly from the Member provided such amount:

- (A) was distributed to the Member by an eligible retirement plan;
- (B) is received by the Plan on or before the 60th day after the day it was received by the Member; and
- (C) would otherwise be includible in gross income, and
- (D) is not attributable to Roth contributions.

Notwithstanding clause (B) above, the Primary Administrator may accept a Rollover Contribution more than 60 days after the amount was received by the Member provided the Member has received from the Secretary of the Treasury a waiver of the 60-day requirement, pursuant to Section 402(c)(3)(B) of the Code.

- (c) Notwithstanding paragraphs (a) and (b) above, effective on and after such date as shall be prescribed by the Primary Administrator, the Plan may accept on behalf of an Employee before-tax amounts that are either:
  - (i) contributed by the Member on or before the 60th day after the day such amounts were received by the Member from an individual retirement account or individual retirement annuity of the Member described in Section 408(a) or 408(b), respectively, of the Code, or
  - (ii) are directly rolled over from such individual retirement account or individual retirement annuity of the Member.

Notwithstanding the foregoing, the Plan shall not accept any amount unless such amount is eligible to be rolled over to a qualified trust in accordance with applicable law and the Member provides evidence satisfactory to the Primary Administrator that such amount qualifies for rollover treatment.

## 3.09 Contributions During Period of Military Leave

(a) Notwithstanding any provision of this Plan to the contrary, contributions and service credit with respect to qualified military service will be provided in accordance with Section 414(u) of the Code. Without regard to any limitations on contributions set forth in this Article 3, a Member who has a period of service in the uniformed services of the United States beginning on or after August 1, 1990 and who returns to service with the Employer having applied to return while his reemployment rights were protected by law may elect to contribute to the Plan the Deferred Cash Contributions, including Catch-Up Contributions that could have been contributed to the Plan in accordance with the

provisions of the Plan had he remained continuously employed by the Employer throughout such period of absence ("make-up contributions"). Effective on and after June 1, 2006, a Member who elects to make Tax- Deferred Contributions and/or Catch-Up Contributions under this paragraph and who would have been able to make Roth Contributions during the period of applicable period of military leave had he been an active member in the Plan during such period, may further elect, pursuant to the provisions of Section 3.01, whether those amounts shall be designated as Tax-Deferred Contributions or Roth Contributions. For purposes of determining the amount of makeup contributions a Member may make, his Compensation for the period of the absence shall be deemed to be the rate of the Member's Compensation (up to the Annual Dollar Limit) he would have received had he remained employed as an Employee for that period or, if such rate is not reasonably certain, on the basis of the Member's rate of annual Compensation during the 12 month period immediately prior to such period of absence (or if shorter, the period of employment immediately preceding such period). Any Deferred Cash Contributions, including Catch-Up Contributions so determined shall be limited as provided in Section 3.01 with respect to the Plan Year or Years to which such contributions relate rather than the Plan Year in which payment is made. Any payment to the Plan described in this paragraph shall be made during the applicable repayment period. The repayment period shall equal three times the period of military leave, but not longer than five years and shall begin on the latest of: (i) the Member's date of reemployment, (ii) October 13, 1996, or (iii) the date the Employer notifies the Employee of his rights under this Section 3.09. The earnings (or losses) on make-up contributions shall be credited commencing with the date the make-up contribution is made in accordance with the provisions of Article 4.

(b) With respect to a Member who makes the election described in paragraph (a) above, the Employer shall make Special Contributions on the make-up contributions in the amount described in the provisions of Section 3.02(b) as in effect for the Plan Year to which such make-up contributions relate. Employer Special Contributions under the preceding sentence shall be made during the period described in paragraph (a) above. In addition, with respect to a Member who is eligible to make the election described in paragraph (a), an Employer shall make any applicable Special Contributions that would have been contributed to the Plan in accordance with the provisions of Section 3.02(a), Section 3.02(c), or Section 3.02(d) had he remained continuously employed by the Employer

throughout the period of absence during which he was in the uniformed services of the United States. The earnings (or losses) on Special Contributions shall be credited commencing with the date the contributions are made in accordance with the provisions of Article 4. Any limitations on Special Contributions shall be applied with respect to the Plan Year or Years to which such contributions relate rather than the Plan Year or Years in which payment is made.

- (c) All contributions under this Section 3.09 (other than Catch-Up Contributions) are considered "annual additions," as defined in Section 415(c)(2) of the Code, and shall be limited in accordance with the provisions of Section 3.06 with respect to the Plan Year or Years to which such contributions relate rather than the Plan Year in which payment is made.
- (d) Notwithstanding the foregoing, effective January 1, 2007, the Plan shall comply with Section 401(a)(37) of the Code and any applicable guidance thereunder.

## ARTICLE 4. INVESTMENT OF CONTRIBUTIONS

#### 4.01 Investment Funds

- (a) Contributions to the Plan shall be invested in one or more Investment Funds, as authorized by the Primary Administrator, which from time to time may include such equity funds, international equity funds, fixed income funds, money market funds, and other funds as the Primary Administrator elects to offer.
- (b) The Trustees may keep such amounts of cash as they, in their sole discretion, shall deem necessary or advisable as part of the Funds, all within the limitations specified in the trust agreement.
- (c) Dividends, interest, and other distributions received on the assets held by the Trustee in respect to each of the above Funds shall be reinvested in the respective Fund.

#### 4.02 Investment of Members' Accounts

A Member shall make one investment election covering the contributions to his Deferred Account, Roth Account, Employer Account, Rollover Account and Transfer Account, in accordance with one of the following options:

- (a) 100% in one of the available Investment Funds;
- (b) in more than one Investment Fund allocated in multiples of 1%, subject to any minimum percentage requirements established by the Primary Administrator, in a single Fund, or such other percentage as the Primary Administrator may permit.

Effective as of October 1, 2003 (or such later date prescribed by the Primary Administrator), a Member may make a separate individual investment election with respect to each of the following types of contributions made to his Accounts:

- (i) Deferred Cash Contributions,
- (ii) Special Contributions,
- (iii) Rollover Contributions, and/or
- (iv) Transfer Contributions.

In the event a Member fails to make a separate investment election with respect to a particular type of contribution, such contribution type shall be invested in accordance with the investment election in effect with respect to the Member's Tax-Deferred Contributions.

Unless a Participant provides, in accordance with administrative procedures established by the Primary Administrator, an initial investment election for the amounts to be re-characterized under an In-Plan Roth Conversion under Sections 7.05 or 9.10, the In-Plan Roth Conversion amount will remain invested among the same Plan investments in which said amounts were invested prior to the In-Plan Roth Conversion. On and after the date an In-Plan Roth Conversion is completed, any transfers/reallocations made pursuant to Section 4.05 shall apply to the Participant's entire Roth Account, including any amount held in the Participant's In-Plan Roth Conversion Subaccount.

## 4.03 Responsibility for Investments

Each Member is solely responsible for the selection of his investment options. The Trustee, the Board, the Primary Administrator, the Employer, and the officers, supervisors and other employees of the Employer are not empowered to advise a Member as to the manner in which his Accounts shall be invested. The fact that an Investment Fund is available to Members for investment under the Plan shall not be construed as a recommendation for investment in that Investment Fund. If a Member fails to make any investment election covering his Accounts, his Accounts shall be invested in the Short Term Investment Fund, or such other Fund as the Primary Administrator determines, under uniform and nondiscriminatory rules, until such time as the Member makes such election.

# 4.04 Change of Election

A Member may change his investment elections under Section 4.02 by giving such advance Notice as the Primary Administrator shall prescribe. The changed investment election shall become effective as soon as administratively practicable following receipt of such Notice, and shall be effective only with respect to subsequent contributions.

#### 4.05 Transfers/Reallocation Between Funds

A Member may elect under uniform rules (effective on and after February 1, 2004, including, but not limited to, rules with respect to market timing) prescribed by the Primary Administrator regarding the reallocation/transfer of all or any portion of a Member's Accounts among the

Investment Funds, in multiples of 1%, by giving such advance Notice as the Primary Administrator shall prescribe. Subject to uniform rules (effective on and after February 1, 2004, including, but not limited to, rules with respect to market timing) prescribed by the Primary Administrator regarding the reallocation/transfer of all or any portion of a Member's Accounts among the Investment Funds, the transfer or reallocation shall be effective as soon as practicable next following receipt of such Notice.

# 4.06 Limitations Imposed by Contract, Prospectus or Other Document of Similar Import

Notwithstanding anything in this Article to the contrary, any contributions invested in a guaranteed investment contract or in an Investment Fund covered by a prospectus or other document of similar import or effect shall be subject to any and all terms of such contract, prospectus or other document of similar import or effect, including any limitations therein placed on the exercise of any rights otherwise granted to a Member under any other provisions of this Plan with respect to such amounts.

## ARTICLE 5. VALUATION OF THE ACCOUNTS

## **5.01** Valuation of the Investment Funds

The Trustees shall value the Investment Funds at least monthly. On each Valuation Date, there shall be allocated to the Accounts of each Member his proportionate share of the increase or decrease in the fair market value of his Accounts in each of the Funds. Whenever an event requires a determination of the value of the Member's Accounts, the value shall be computed as of the Valuation Date coincident with or immediately following the date of determination, subject to the provisions of Section 5.02.

# **5.02** Discretionary Power of the Primary Administrator

The Primary Administrator reserves the right to change from time to time the procedures used in valuing the Accounts or crediting (or debiting) the Accounts if it determines, after due deliberation and upon the advice of counsel and/or the current recordkeeper, that such an action is justified in that it results in a more accurate reflection of the fair market value of assets. In the event of a conflict between the provisions of this Article and such new administrative procedures, those new administrative procedures shall prevail.

#### 5.03 Statement of Accounts

At least once a year, each Member shall be furnished with a statement setting forth the value of his Accounts by type of Investment Fund and a schedule of transactions during the statement period.

# **5.04** Valuation Dates

The Primary Administrator shall establish procedures for determining the Valuation Dates which shall apply for withdrawals, distributions, or other relevant purposes. Valuation Dates need not be the same for all purposes.

# ARTICLE 6. VESTED PORTION OF ACCOUNTS

# 6.01 Vesting

A Member shall at all times be 100% vested in, and have a nonforfeitable right to, his Deferred Account, his Roth Account, his Employer Account, his Rollover Account and his Transfer Account insofar as such is not contrary to law or applicable Code requirements.

#### ARTICLE 7. WITHDRAWALS WHILE STILL EMPLOYED

#### 7.01 Withdrawal After Age 59½

A Member who shall have attained age 59½ as of the effective date of any withdrawal pursuant to this Section may, subject to Section 7.03, elect to withdraw, in the following order, all or part of his Employer Account, his Transfer Account, his Rollover Account and his Deferred Account: Effective as of June 1, 2006, a Member who has withdrawn all of his Employer Account, Transfer Account and Rollover Account may, subject to Section 7.03, make a separate election under this Section 7.01 to withdraw next either all or part of his Roth Account, or all or part of Deferred Account, or any combination thereof.

#### 7.02 Hardship Withdrawal

- (a) A Member may, subject to the provisions of Section 7.03, elect to withdraw in the following order all or part of (i) his Rollover Account, (ii) his Transfer Account, (iii) his Employer Account, (iv) effective as of April 1 2008, his Roth Contributions, and then (v) his Tax-Deferred Contributions (including any earnings credited to his Deferred Account prior to January 1, 1989), provided that he furnishes proof of "Hardship" satisfactory to the Primary Administrator in accordance with the provisions of paragraphs (b) and (c) below. Notwithstanding the foregoing, amounts held in a Member's Employer Account cannot be withdrawn on or after June 1, 2008 under the provisions of this Section 7.02.
- (b) A Member shall be considered to have incurred a "Hardship" if, and only if, he meets the requirements of paragraphs (c) and (d) below or of Section 7.06.
- (c) As a condition for Hardship there must exist with respect to the Member an immediate and heavy need to draw upon his Accounts. The Primary Administrator shall presume the existence of such immediate and heavy need if the requested withdrawal is on account of any of the following:
  - (i) expenses for medical care described in Section 213(d) of the Code previously incurred by the Member, his spouse or any of his dependents (as defined in Section 152 of the Code) or necessary for these persons to obtain such medical care;

- (ii) costs directly related to the purchase of a principal residence of the Member (excluding mortgage payments);
- (iii) payment of tuition, related educational fees, and expenses for room and board for the next 12 months of post-secondary education of the Member, his spouse or dependents;
- (iv) payment of amounts necessary to prevent eviction of the Member from his principal residence or to avoid foreclosure on the mortgage of his principal residence; or
- (v) payments for burial or funeral expenses for the Member's deceased parent, spouse, children or dependents (as defined in Section 152 of the Code and without regard to Section 152(d)(1)(B) of the Code);
- (vi) expenses for the repair of damages to the Member's principal residence that would qualify for the casualty deduction under Section 165 of the Code (determined without regard to whether the loss exceeds 10 percent of the Member's adjusted gross income); or
- (vii) the inability of the Member to meet such other expenses, debts or other obligations recognized by the Internal Revenue Service as giving rise to immediate and heavy financial need for purposes of Section 401(k) of the Code.

The amount of the withdrawal may not be in excess of the amount of the immediate and heavy financial need of the employee including any amounts necessary to pay any federal, state, or local income taxes and any amounts necessary to pay any penalties reasonably anticipated to result from the distribution.

In evaluating the relevant facts and circumstances, the Primary Administrator shall act in a nondiscriminatory fashion and shall treat uniformly those Members who are similarly situated. The Member shall furnish to the Primary Administrator such supporting documents as the Primary Administrator may request in accordance with uniform and nondiscriminatory rules prescribed by the Primary Administrator.

(d) As a condition for a Hardship withdrawal, the Member must demonstrate that the requested withdrawal is necessary to satisfy the financial need described in paragraph (c). To demonstrate such necessity, the Member who requests a hardship withdrawal to

satisfy a financial need described in (c) above must comply with either (i) or (ii) as follows:

- (i) The Member must certify to the Primary Administrator, on such form as the Primary Administrator may prescribe, that the financial need cannot be fully relieved (A) through reimbursement or compensation by insurance or otherwise, (B) by reasonable liquidation of the Member's assets, (C) by cessation of Deferred Cash Contributions, or (D) by other distributions or nontaxable (at the time of the loan) loans from the Plan or other plans of the Employer or by borrowing from commercial sources at a reasonable rate in an amount sufficient to satisfy the need. The actions listed are required to be taken to the extent necessary to relieve the hardship but any action which would have the effect of increasing the hardship need not be taken. For purposes of this clause (i) there shall be attributed to the Member those assets of the Member's spouse and minor children which are reasonably available to the Member. The Member shall furnish to the Primary Administrator such supporting documents as the Primary Administrator may request in accordance with uniform and nondiscriminatory rules prescribed by the Primary Administrator. If, on the basis of the Member's certification and the supporting documents, the Primary Administrator finds it can reasonably rely on the Member's certification, then the Primary Administrator shall find that the requested withdrawal is necessary to meet the Member's financial need.
- (ii) The Member must request, on such form as the Primary Administrator may prescribe, that the Primary Administrator makes its determination of the necessity for the withdrawal solely on the basis of his application. In that event, the Primary Administrator shall make such determination, provided all of the following requirements are met: (A) the Member has obtained all distributions, other than distributions available only on account of Hardship, and all nontaxable loans currently available under all plans of the Employer and (B) if the Hardship withdrawal will be made from a Member's Deferred Account (or, effective as of April 1, 2008, Roth Account), (1) the Member is prohibited from making contributions (other than Rollover or Transfer Contributions) to the Plan and all other plans of the Employer under the terms of such plans or by means of an otherwise legally enforceable agreement for at least six months after receipt of the distribution. Effective for hardship withdrawals received in 2001, the

Member is prohibited from making contributions (other than Rollover or Transfer Contributions) to the Plan and all other plans of the Employer under the terms of such plans or by means of an otherwise legally enforceable agreement until the later of January 1, 2002 or the date that is six months after receipt of the distribution. For purposes of this subparagraph (ii), "all other plans of the Employer" shall include qualified and non-qualified deferred compensation plans and such other plans as may be designated under regulations issued under Section 401(k) of the Code, but shall not include health and welfare benefit plans or the mandatory employee contribution portion of a defined benefit plan.

#### 7.03 Procedures and Restrictions

To make a withdrawal, a Member shall give at least 30 days', or such shorter period as the Primary Administrator determines, prior Notice to the Primary Administrator. A withdrawal shall be made as of the Valuation Date next following the expiration of the Notice period. The minimum withdrawal shall be \$500, or the total value of his Accounts available for withdrawal, if less. If a loan and a hardship withdrawal are processed as of the same Valuation Date, the amount available for the hardship withdrawal will equal the vested portion of the Member's Accounts on such valuation date reduced by the amount of the loan. The amount of the withdrawal shall be allocated between and among the Investment Funds in proportion to the value of the Member's Accounts from which the withdrawal is made in each Investment Fund as of the date of the withdrawal. All payments to Members under this Article shall be made in cash as soon as practicable.

## 7.04 Transfers directly to the Teachers' and State Employees' Retirement System of North Carolina, the Consolidated Judicial Retirement System of North Carolina, the Legislative Retirement System of North Carolina, the North Carolina Local Government Employees' Retirement System and the Charlotte Firefighters Retirement System.

A Member may voluntary elect to transfer all or a portion of his Accounts (excluding his Roth Account) to the Teachers' and State Employees' Retirement System of North Carolina, the Consolidated Judicial Retirement System of North Carolina, the Legislative Retirement System of North Carolina, the North Carolina Local Government Employees' Retirement System or the Charlotte Firefighters Retirement System in order to purchase prior service credits or permissive service credits under such applicable Retirement System.

Any amount transferred pursuant to the provisions of this Section 7.04, shall be subject to the provisions of the applicable Retirement System and held in the annuity savings fund or other member's contribution fund of the trust thereunder. The trustee of this Plan shall transfer such amounts to the trustee of the applicable Retirement System. A Member's election to have such transfer made on his behalf shall be made in accordance with the procedures established by the Primary Administrator. The amount of the transfer shall be allocated between and among the Investment Funds under this Plan as of the date of the transfer. All transfers under this Section shall be made in cash, as soon as practicable following receipt of such advance Notice as the Primary Administrator shall prescribe.

#### 7.05 In-Plan Roth Conversions

Effective as of December 1, 2010, a Participant who is entitled to a distribution under the terms of Section 7.01 may elect instead to re-characterize the amount of a distribution (except amounts attributable to Roth contributions or Roth rollover contributions) which he is requesting as an In-Plan Roth Conversion. For the avoidance of doubt, the funds held in a Participant's Roth Account are not eligible for an In-Plan Roth Conversion under the provisions of this Section 7.05. Such re-characterized amounts shall be held in an In-Plan Roth Conversion Subaccount that has been, or that will be, established to hold such funds. For purposes of this Section 7.05, the restrictions of Section 7.03 shall not apply.

#### 7.06 Hurricane Sandy Relief

(a) To the extent that a hardship withdrawal described in Section 7.02(a) is elected by a Sandy Member, as defined in Section 7.06(b), the hardship withdrawal provisions of the Plan set forth in Sections 7.02(b), 7.02(c), and 7.02(d) are modified to adopt the liberalized hardship withdrawal standards and procedural requirements set forth in Internal Revenue Service Announcement 2012-44 ("Announcement 2012-44"). For purposes of clarity, a hardship withdrawal elected by a Sandy Member shall otherwise continue to be subject to the requirements of Section 7.02(a). Any such hardship withdrawal will be treated as a hardship withdrawal for all purposes under the Code except as otherwise provided herein or by law.

#### (b) A "Sandy Member" is a Member whose:

 (i) Principal residence on October 26, 2012 was located in one of the counties or Tribal Nations that have been identified as covered disaster areas because of the devastation caused by Hurricane Sandy;

- (ii) Place of employment was located in one of these counties or Tribal Nations; or
- (iii) Lineal ascendant or descendant, dependent or spouse had a principal residence or place of employment in one of these counties or Tribal Nations on October 26, 2012.
- (c) The Board will make a good-faith effort to comply with the Plan's procedural requirements for hardship withdrawals made by Sandy Members and will make a reasonable attempt to assemble any missing documentation as soon as practicable, and otherwise act consistent with Announcement 2012-44.
- (d) This Section 7.06 applies to Hardship distributions made to Sandy Members on or after October 26, 2012 and no later than February 1, 2013.

#### **ARTICLE 8. LOANS TO MEMBERS**

#### **8.01** Amount Available

- (a) Subject to the following provisions of this Article 8, a Member who is an Employee of an Employer may borrow, on Notice to the Primary Administrator and on approval by the Primary Administrator under such uniform rules as it shall adopt, up to 90% (100% effective with respect to loans made on or after October 1, 2003, or such later date prescribed by the Primary Administrator) of his Accounts. Notwithstanding the foregoing, such Member may borrow no more than an amount which equals the lesser of (i) or (ii) where (i) is \$50,000, and (ii) is the greater of \$10,000 or one-half of the value of his Accounts (with respect to a loan made on or after June 1, 2006 and prior to April 1, 2008, excluding his Roth Account). The minimum loan shall be \$1,000. All loans previously made shall be subject to the rules in effect under the Plan at the time the loan was made; however, any loans made, renewed, renegotiated, modified or extended on or after January 1, 1987 shall be subject to the provisions of this Article 8 as in effect at that time.
- (b) Loans from the Plan shall be repaid with interest. The interest rate or rates to be charged on loans made during the Plan Year shall be reasonable as determined by the Primary Administrator, and shall take into account interest rates currently in effect. The Primary Administrator shall not discriminate among Members in the matter of interest rates, but loans may bear different interest rates if granted or outstanding at different times, if the difference is justified by a change in general economic conditions or the terms of the particular loan.
- (c) The amount of the loan is to be transferred to a special "Loan Fund" for the Member under the Plan. The Loan Fund consists solely of the amount of the Member's Accounts transferred to the Loan Fund and is invested solely in the loan made to the Member. The amount of the Member's Accounts transferred to the Loan Fund shall be pledged as security for the loan. No such amount shall be available for withdrawal under Article 7. Payments of principal on the loan will reduce the amount held in the Member's Loan Fund. Those payments, together with the attendant interest payment, will be credited to the Member's Accounts and invested in the Investment Funds in accordance with the Member's then effective investment election.

#### **8.02** Terms

- (a) In addition to such rules and regulations as the Primary Administrator may adopt, all loans shall comply with the following terms and conditions:
  - (i) An application for a loan by a Member shall be made with prior Notice to the Primary Administrator, whose action in approving or disapproving the application shall be final. Effective on and after January 1, 2004, the Member shall certify in such application as to the existence and amount of any outstanding loans (including any loans deemed distributed) from any qualified plan maintained by the Employer).
  - (ii) Each loan shall be evidenced by a promissory note payable to the Plan;
  - (iii) The period of repayment for any loan shall be arrived at by mutual agreement between the Primary Administrator and the Member, subject to any rules and restrictions established by the Primary Administrator pursuant to paragraph (c) below.
  - (iv) If a Member with an outstanding loan takes an authorized leave of absence without pay or reduced pay that is less than the required loan payments, for reasons other than to enter the uniformed services of the United States, loan payments may be suspended at the request of the Member, for a period of up to 12 months or until the end of the term of the loan, if earlier. Upon a Member's reemployment from the leave of absence, the Member shall resume payments either in the same amount as before the leave with the full balance due upon the expiration of the repayment period or by re-amortizing the loan in substantially level installments over the remaining term of the loan.
  - (v) If a Member takes a leave of absence to enter the uniformed services of the United States, loan repayments shall be suspended during the period of leave. Upon the Member's reemployment from the uniformed services, the period of repayment shall be extended by the number of months of the period of service in the uniformed services or, if greater, the number of months that would remain if the original loan term were five years plus the number of months in the period of

absence; provided, however, if the Member incurs a termination of employment and requests a distribution pursuant to Article 9, the loan shall be canceled, and the outstanding loan balance shall be distributed pursuant to Article 9. If a Member enters the uniformed services of the United States the interest rate applicable to the unpaid loan balance during the period of leave shall be reduced to 6%, in accordance with the Soldiers' and Sailors' Civil Relief Act of 1940. Upon a Member's reemployment from the leave of absence, the Member shall resume payments either in the same amount as before the leave with the full balance due upon the expiration of the repayment period or by re-amortizing the loan in substantially level installments over the remaining term of the loan.

- (vi) Payments of principal and interest will be made by payroll deductions or in a manner agreed to by the Member and the Primary Administrator in substantially level amounts, but no less frequently than quarterly, in an amount sufficient to amortize the loan over the repayment period;
- (vii) A loan may be prepaid in full as of any date without penalty;
- (viii) Only one loan may be outstanding at any given time.
- (ix) Effective January 1, 2004, if at the time a loan is to be issued to a Member a prior loan has been deemed distributed to the Member and not repaid, a new loan may only be issued to a Member if the Member enters into an agreement, enforceable under law that requires repayment by payroll withholding.
- (b) If a loan is not repaid in accordance with the terms contained in the promissory note and a default occurs, the Plan may execute upon its security interest in the Member's Accounts under the Plan to satisfy the debt; however, the Plan shall not levy against any portion of the Loan Fund attributable to amounts held in the Member's Accounts until such time as a distribution of the Member' Accounts could otherwise be made under the Plan.

(c) Any additional rules or restrictions as may be necessary to implement and administer the loan program shall be in writing and communicated to employees. Such further documentation is hereby incorporated into the Plan by reference, and the Primary Administrator is hereby authorized to make such revisions to these rules as it deems necessary or appropriate, on the advice of counsel.

#### ARTICLE 9. DISTRIBUTION OF ACCOUNTS UPON TERMINATION OF EMPLOYMENT

#### 9.01 Eligibility

Upon a Member's termination of employment or death the entire amount to the credit of his Accounts shall be distributed as provided in this Article.

#### 9.02 Forms of Distribution

- (a) Unless the Member (or his sole Beneficiary in the event of the Member's death prior to the date benefit payments commence) elects otherwise, distribution of the Member's Accounts shall be made in a cash lump sum. A Member (or his sole Beneficiary in the event of the Member's death prior to the date benefit payments commence) may elect to receive an optional form of payment described below:
  - (i) Payments in approximately equal monthly installments over a period designated by the Member not to exceed the Member's (or the sole Beneficiary's) life expectancy.
  - (ii) Payments in approximately equal quarterly installments over a period designated by the Member not to exceed the Member's (or the sole Beneficiary's) life expectancy. (iii) Payments in approximately equal semi-annual installments over a period designated by the Member not to exceed the Member's (or the sole Beneficiary's) life expectancy.
  - (iv) Payments in approximately equal annual installments over a period designated by the Member not to exceed the Member's (or the sole Beneficiary's) life expectancy.
  - (v) Payments in approximately equal monthly, quarterly, semi-annual, or annual installments over the applicable distribution period determined in accordance with the Uniform Lifetime Table set forth in Section 1.401(a)(9)-9 of the Treasury regulations under Section 401(a)(9) of the Code.

A Member (or the sole Beneficiary) who elects installment payments may, at any time after such payments commence, elect to receive the remaining value of any unpaid installments in a lump sum by filing a request with the Primary Administrator on such form or forms as it may prescribe. In the event a Member dies before all installments have been paid, and his surviving Beneficiary is a sole person; such installments shall continue to the Beneficiary, unless such Beneficiary elects otherwise. Notwithstanding

the foregoing, if the Member dies before all installments have been paid and his Beneficiary is other than a sole person, the remaining balance in his Accounts shall be paid in an immediate cash lump sum to such Beneficiary. In the event a sole Beneficiary who has elected installment payments under the preceding provisions of this paragraph (a) dies before all installments have been paid, the remaining balance in the Member's Accounts shall be paid in a lump sum to the estate of such Beneficiary. Notwithstanding the foregoing, a Member who has elected to receive installment payments over his life expectancy under clauses (i), (ii), (iii), or (iv) above and who has attained age 70½ prior to January 1, 2002, may elect to revise his form of distribution and have his future payments made pursuant to the provision of clause (v) above.

- (b) Notwithstanding the preceding, if a Member dies before his benefits commence and his Beneficiary is other than a sole person, the value of his Accounts shall be paid to his Beneficiary in a lump sum.
- (c) In the event a Member is required to begin receiving payments in accordance with Section 9.03(d) and the Member has not elected a form of payment described in paragraph (a) above, the Employer shall distribute to the Member in each distribution calendar year the minimum amount necessary to satisfy the minimum distribution requirements provisions of Section 401(a)(9) of the Code; provided, however, that the payment for the first distribution calendar year shall be made on or before April 1 of the following calendar year.

Notwithstanding the foregoing, with respect to distribution calendar years commencing on and after January 1, 2003, such minimum amount shall be lesser of:

- (i) the quotient obtained by dividing the Member's Accounts by the distribution period in the Uniform Lifetime Table set forth in Section 1.401(a)(9)-9 of the Treasury regulations, using the Member's age as of the Member's birthday in the distribution calendar year; or
- (ii) if the Member's sole designated beneficiary for the distribution calendar year is the Member's spouse, the quotient obtained by dividing the Member's Accounts by the number in the Joint and Last Survivor Table set forth in Section 1.401(a)(9)-9 of the Treasury regulations, using the Member's and spouse's

attained ages as of the Member's and the spouse's birthdays in the distribution calendar year.

For purposes of this paragraph (c), the following definitions apply:

- (i) "Designated beneficiary" means the individual who is designated as the Beneficiary and is the designated beneficiary under Section 401(a)(9) of the Code and Section 1.401(a)(9)-4 of the Treasury regulations.
- (ii) "Distribution calendar year" means a calendar year for which a minimum distribution is required. The first distribution calendar year is the calendar year in which the Member who has terminated employment attains age 70½.
- (iii) "Life expectancy" means life expectancy as computed by use of the Single Life Table in Section 1.401(a)(9)-9 of the Treasury regulations.
- (iv) "Member's Accounts" means the balance of the Member's Accounts as of the last Valuation Date in the calendar year immediately preceding the distribution calendar year ("valuation calendar year") increased by the amount of contributions made and allocated or forfeitures allocated to the Member's Accounts as of dates in the valuation calendar year after such last Valuation Date and decreased by distributions made in the valuation calendar year after such last Valuation Date. The Member's Accounts for the valuation calendar year includes any amounts rolled over or transferred to the Plan either in the valuation calendar year or in the distribution calendar year if distributed or transferred in the valuation calendar year.
- (d) Effective January 1, 2009, notwithstanding any provision of this Article 9 to the contrary, a Member or Beneficiary who would have been required to receive payments in accordance with Section 9.03(d) below for 2009 but for the enactment of Code section 401(a)(9)(H) ("2009 RMDs"), and who would have satisfied that requirement by receiving distributions that are equal to the 2009 RMDs will: (1) for Members or Beneficiaries who received payments in accordance with Section 9.03(d) prior to 2009, receive 2009 RMDs unless the Member or Beneficiaries who had not received payments in accordance with Section 9.03(d) prior to 2009, not receive 2009 RMDs unless the Member or Beneficiary affirmatively elects to receive such distributions. In addition, notwithstanding any provision of this Article 9 to the contrary, and solely for

purposes of applying Section 9.08, 2009 RMDs will not be eligible for direct rollover from the Plan.

#### 9.03 Commencement of Payments

- (a) Except as otherwise provided in this Article, distribution of a Member's Accounts shall commence as soon as administratively practicable following the later of (i) the Member's termination of employment or (ii) the 65th anniversary of the Member's date of birth (but not more than 60 days after the close of the Plan Year in which the later of (i) or (ii) occurs).
- (b) In lieu of a distribution as described in paragraph (a) above, a Member may elect to have the distribution of his Accounts commence as of any Valuation Date coincident with or following his termination of employment, provided that such date is not later than the date described in paragraph (d) below.
- (c) Notwithstanding the above provisions of Article 9, if a Member dies in active service or terminates employment and dies before distribution of his Accounts has commenced, distribution of the Member's Accounts to his designated Beneficiary shall not commence later than one year after the Member's death; provided, however, if the designated Beneficiary is the Member's surviving spouse, distribution shall commence not later than the date on which the Member would have attained age 70½.
- (d) In no event, however, shall the provisions of this Section operate so as to allow the distribution of a Member's Accounts to begin later than the April 1 following the calendar year in which he attains age 70½ or retires, if later.

#### 9.04 Mandatory Distribution of De Minimis Accounts

(a) Notwithstanding any provision of the Plan to the contrary, if a Member has a termination of employment and the value of the Member's Accounts amounts to \$1,000 or less, the Primary Administrator shall distribute the Member's entire account in a lump sum payment as soon as administratively practicable following notification to the Plan's recordkeeper of the Member's termination of employment; provided that the Member is not reemployed by an Employer as of the date of distribution and the value of the Member's Accounts remains less than \$1,000 as of the date of the distribution.

(b) Notwithstanding the above, in the event that the Member's account is deemed to be Bailey Vested, the Member's account will not be subject to a mandatory distribution under Section 9.04(a). For the purposes of this section 9.04(b), "Bailey Vested" shall mean a Member who had contributed or contracted to contribute to the Plan prior to August 12, 1989.

#### 9.05 Status of Accounts Pending Distribution

Until completely distributed under Section 9.03, the Accounts of a Member who is entitled to a distribution at a future date shall continue to be invested as part of the funds of the Plan and the Member (or his Beneficiary in the event of the death of the Member) shall retain investment transfer rights as described in Section 4.05 during this period. Until the commencement of payments under Section 9.03, a Member shall be permitted to make a partial withdrawal from his Account pursuant to the provisions of Section 7.03.

#### 9.06 Proof of Death and Right of Beneficiary or Other Person

The Primary Administrator may require and rely upon such proof of death and such evidence of the right of any Beneficiary or other person to receive the value of the Accounts of a deceased Member as the Primary Administrator may deem proper and its determination of the right of that Beneficiary or other person to receive payment shall be conclusive.

#### 9.07 Distribution Limitation

Notwithstanding any other provision of this Article 9, all distributions from the Plan shall conform to the requirements of Section 401(a)(9) of the Code, including the incidental death benefit provisions of Section 401(a)(9)(G) of the Code. Such requirements shall be administered in accordance with the regulations issued under Section 401(a)(9) of the Code, as follows:

- (a) With respect to distributions made for distribution calendar years beginning on or after January 1, 2001, the Plan will apply the minimum distribution requirements of Section 401(a)(9) of the Code in accordance with the regulations under Section 401(a)(9) that were proposed on January 17, 2001.
- (b) With respect to distributions made for distribution calendar years beginning on and after January 1, 2003, the Plan will apply the minimum distribution requirements of Section 401(a)(9) of the Code in accordance with the regulations under Section 401(a)(9) that were issued April 17, 2002, as prescribed in Section 9.02.

The provisions of Section 401(a)(9) of the Code and the regulations thereunder shall override any Plan provision that is inconsistent with Section 401(a)(9) of the Code.

#### 9.08 Direct Rollover of Certain Distributions

Notwithstanding any provision of the Plan to the contrary that would otherwise limit a distributee's election under this Section, a distributee may elect, at the time and in the manner prescribed by the Primary Administrator, to have any portion of an eligible rollover distribution paid directly by the Plan to an eligible retirement plan specified by the distributee in a direct rollover. The following definitions apply to the terms used in this Section:

- "Eligible rollover distribution" means any distribution of all or any portion of the (a) balance to the credit of the distributee, except that an eligible rollover distribution does not include any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the distributee or the joint lives (or joint life expectancies) of the distributee and the distributee's designated beneficiary, or for a specified period of ten years or more, any distribution to the extent such distribution is required under Section 401(a)(9) of the Code, after-tax amounts unless such amount is rolled over to an individual retirement account or individual retirement annuity described in Section 408(a) or 408(b) of the Code, respectively, or to a defined contribution plan qualified under Section 401(a) of the Code that agrees to separately account for such amount; effective on and after January 1, 2002, any in-service withdrawal made on account of hardship and effective as of June 1, 2006, any distribution from the Roth Account unless such amount is rolled over or transferred (i.e., directly rolled over) to a Roth IRA (as defined in Section 408A(b) of the Code), or transferred to a defined contribution plan qualified under Section 401(a) of the Code that agrees to separately account for such amount, or, solely with respect to the amount that would otherwise be included in gross income, is rolled over to a defined contribution plan qualified under Section 401(a) of the Code that agrees to separately account for such amount.
- (b) "Eligible retirement plan" means any of the following types of plans that accept the distributee's eligible rollover distribution:
  - (i) an individual retirement account or individual retirement annuity described in Section 408(a) or 408(b) of the Code, respectively;

- (ii) a qualified trust described in Section 401(a) of the Code;
- (iii) effective January 1, 2002, an annuity contract described in Section 403(b) of the Code;
- (iv) effective January 1, 2002, an eligible plan under Section 457(b) of the Code which is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state and which agrees to separately account for amounts transferred into such plan from this Plan, and
- (v) a Roth IRA (as defined in Section 408A(b) of the Code) solely with respect to Roth Contributions.
- (c) "Distributee" means an employee or former employee. In addition, the employee's or former employee's surviving spouse and the employee's or former employee's spouse or former spouse who is the alternate payee under a qualified domestic relations order as defined in Section 414(p) of the Code, are distributees with regard to the interest of the spouse or former spouse; and
- (d) "Direct rollover" means a payment by the Plan to the eligible retirement plan specified by the distributee.

Notwithstanding any provision of this Section to the contrary, effective as of January 1, 2007, the non-spouse Beneficiary of a deceased Member may elect, at the time and in the manner prescribed by the Committee, to directly rollover any portion of his distribution from the Plan to an individual retirement account or annuity described in Section 408(a) or 408(b) of the Code ("IRA") that is established on behalf of the designated Beneficiary and that will be treated as an inherited IRA pursuant to the provisions of Section 402(c)(11) of the Code.

In the event that the provisions of this Section 9.08 or any part thereof cease to be required by law as a result of subsequent legislation or otherwise, this Section 9.08 or any applicable part thereof shall be ineffective without the necessity of further amendment to the Plan.

#### 9.09 Waiver of Notice Period

Except as provided in the following sentence, an election by the Member to receive a distribution prior to age 65 shall not be valid unless the election is made (a) after the Member has received the notice required under Section 1.411(a)-11(c) of the Income Tax Regulations and (b) within a reasonable time before the effective date of the commencement of the distribution as prescribed

by said regulations. If such distribution is one to which Section 401(a)(11) and 417 of the Code do not apply, such distribution may commence less than 30 days after the notice required under Section 1.411(a)-11(c) of the Income Tax Regulations is given, provided that:

- (i) the Board clearly informs the Member that he has a right to a period of at least 30 days after receiving the notice to consider the decision of whether or not to elect a distribution (and, if applicable, a particular distribution option), and
- (ii) the Member, after receiving the notice, affirmatively elects a distribution.

#### 9.10 In-Plan Roth Conversions

Effective as of December 1, 2010, a Participant who is entitled to a distribution under the term of this Article 9 may elect instead to recharacterize the amount of a distribution (except amounts attributable to Roth contributions or Roth rollover contributions which he is requesting as an In-Plan Roth Conversion. For the avoidance of doubt, the funds held in a Participant's Roth Account are not eligible for an In-Plan Roth Conversion under the provisions of this Section 9.10. Such recharacterized amounts shall be held in an In-Plan Roth Conversion Subaccount that has been, or that will be, established to hold such funds. For the period from December 1, 2010 through October 1, 2011 only, this Section 9.10 shall apply as well to a spouse Beneficiary of a Participant.

#### ARTICLE 10. ADMINISTRATION OF PLAN

#### 10.01 Appointment of Primary Administrator

The general administration of the Plan and the responsibility for carrying out the provisions of the Plan shall be placed with the Retirement Systems Division of the Department of State Treasurer.

#### 10.02 Duties of Primary Administrator

The Primary Administrator in carrying out the provisions of the Plan and in conjunction with the Board, may retain a Third Party Administrator to carry out the provisions of the Plan; and may delegate to the Third Party Administrator certain duties under the Plan. .

#### 10.03 Individual Accounts

The Third Party Administrator shall maintain, or cause to be maintained, records showing the individual balances in each Member's Accounts. However, maintenance of those records and Accounts shall not require any segregation of the funds of the Plan.

#### 10.04 Meetings

The Primary Administrator shall hold meetings upon such notice, at such place or places, and at such time or times as it may from time to time determine.

#### 10.05 Establishment of Rules

Subject to the limitations of the Plan, the Primary Administrator from time to time shall establish rules for the administration of the Plan and the transaction of its business. The Primary Administrator, in conjunction with the Board, shall have the total and complete discretionary authority to construe and interpret the Plan, to make factual determinations and to determine all questions arising in the administration, interpretation, and application of the Plan (including, but not limited to, the power to make a determination of an individual's eligibility for Plan participation, the right, amount, form, and timing of any benefit payable under the Plan and the date on which any individual ceases to be a Member). The Primary Administrator's discretionary authority shall also include the power to make such adjustments which it deems necessary or desirable to correct any mathematical or accounting errors, and to make a determination of whether a domestic relations order constitutes a Qualified Domestic Relations Order and whether the Alternate Payee otherwise qualifies for benefits hereunder. The Primary Administrator may correct any defect, supply any omission, or reconcile any inconsistency in such manner and to

such extent as it shall deem necessary to carry out the purposes of the Plan. The determination of the Primary Administrator as to the interpretation of the Plan or any disputed question shall be binding, conclusive, and final as to all parties to the extent permitted by applicable law.

#### 10.06 Prudent Conduct

The Primary Administrator, the Board and the Third Party Administrator shall use that degree of care, skill, prudence and diligence that a prudent man acting in a like capacity and familiar with such matters would use in his conduct of a similar situation.

#### 10.07 Appointment of Investment Manager

The Primary Administrator in conjunction with the Board may, in its discretion, appoint one or more investment managers to manage (including the power to acquire and dispose of) all or part of the assets of the Plan, as the Employer shall designate. In that event authority over and responsibility for the management of the assets so designated shall be the sole responsibility of that investment manager.

#### **ARTICLE 11. MANAGEMENT OF FUNDS**

#### 11.01 Trust Agreement

All the funds of the Plan shall be held by the Board in its separate capacity as trustee under a trust agreement in the form the North Carolina General Statutes adopted, or as amended, by the General Assembly of North Carolina for use in providing the benefits of the Plan and paying its expenses not paid directly by the Employer. The Employer shall have no liability for the payment of benefits under the Plan or for the administration of the funds paid over to the Plan.

#### 11.02 Exclusive Benefit Rule

Except as otherwise provided in the Plan, no part of the corpus or income of the funds of the Plan shall be used for, or diverted to, purposes other than for the exclusive benefit of Members and other persons entitled to benefits under the Plan. No person shall have any interest in or right to any part of the earnings of the funds of the Plan, or any right in, or to, any part of the assets held under the Plan, except as and to the extent expressly provided in the Plan not in conflict with the North Carolina General Statutes adopted, or as amended, by the General Assembly of North Carolina.

#### 11.03 Investment In Group Trust

Notwithstanding any provision of the Plan to the contrary, the Plan's assets may be invested in a group trust described in Revenue Ruling 81-100 or a successor thereto, and the terms of such group trust shall be deemed incorporated into this Plan document.

#### **ARTICLE 12. GENERAL PROVISIONS**

#### 12.01 Nonalienation

- (a) Except as required by any applicable law, no benefit under the Plan shall in any manner be anticipated, assigned or alienated, and any attempt to do so shall be void. However, payment shall be made in accordance with the provisions of any judgment, decree, or order which:
  - (i) creates for, or assigns to, a spouse, former spouse, or child of a Member the right to receive all or a portion of the Member's benefits under the Plan for the purpose of providing child support or marital property rights to that spouse, former spouse, or child,
  - (ii) is made pursuant to a state domestic relations law,
  - (iii) does not require the Plan to provide any type of benefit, or any option, not otherwise provided under the Plan, and
  - (iv) otherwise meets the requirements of Section 414(p)(1)(A)(i) of the Internal Revenue Code, as a "qualified domestic relations order", as determined by the Primary Administrator.
- (b) Notwithstanding anything herein to the contrary, if the amount payable to the spouse, former spouse, or child under the qualified domestic relations order is \$5,000 or less, such amount shall be paid in one lump sum as soon as practicable following the determination that said order is a qualified domestic relations order. If the amount exceeds \$5,000, it may be made as soon as practicable following the qualification of the order if the domestic relations order so provides; otherwise it may not be payable before the earliest of (i) the Member's termination of employment, (ii) the time such amount could be withdrawn under Article 7, or (iii) the Member's attainment of age 50.

#### 12.02 Conditions of Employment Not Affected by Plan

The establishment of the Plan shall not confer any legal rights upon any Employee or other person for a continuation of employment, nor shall it interfere with the rights of the Employer to discharge any Employee and to treat him without regard to the effect which that treatment might have upon him as a Member or potential Member of the Plan.

#### 12.03 Facility of Payment

If the Primary Administrator shall find that a Member or other person entitled to a benefit is unable to care for his affairs because of illness or accident or is a minor, the Primary Administrator may direct that any benefit due him, unless claim shall have been made for the benefit by a duly appointed legal representative, be paid to his spouse, a child, a parent or other blood relative, or to a person with whom he resides. Any payment so made shall be a complete discharge of the liabilities of the Plan for that benefit.

#### 12.04 Information

Each Member, Beneficiary or other person entitled to a benefit, before any benefit shall be payable to him or on his account under the Plan, shall file with the Primary Administrator the information that it shall require to establish his rights and benefits under the Plan.

#### 12.05 Construction

- (a) The Plan shall be construed, regulated and administered under the laws of the State of North Carolina and the Code.
- (b) The masculine pronoun shall mean the feminine wherever appropriate.
- (c) The titles and headings of the Articles and Sections in this Plan are for convenience only. In the case of ambiguity or inconsistency, the text rather than the titles or headings shall control.

#### 12.06 Erroneous Allocation

Notwithstanding any provision of the Plan to the contrary, if a Member's Account is credited with an erroneous amount due to a mistake in fact or law, the Primary Administrator shall adjust such Account in such equitable manner as it deems appropriate to correct the erroneous allocation.

#### 12.07 Elections

Any elections, notifications or designations made by a Member pursuant to the provisions of the Plan shall be made in a time and manner determined by the Primary Administrator under rules uniformly applicable to all employees similarly situated. The Primary Administrator reserves the right to change from time to time the time and manner for making notifications, elections or designations by Members under the Plan if it determines after due deliberation that such action is

justified in that it improves the administration of the Plan. In the event of a conflict between the provisions for making an election, notification or designation set forth in the Plan and such new administrative procedures, those new administrative procedures shall prevail.

#### 12.08 Plan Expenses

Unless paid by the Employer, reasonable plan expenses shall be paid from the Trust in accordance with such uniform and nondiscriminatory rules as shall be prescribed by the Primary Administrator. Such rules may include a specific allocation of expenses to a particular investment fund or a particular Member's Account or to a particular Member if the Primary Administrator determines that such allocation of expense is desirable for the equitable administration of the Plan.

#### ARTICLE 13. AMENDMENT, MERGER AND TERMINATION

#### 13.01 Amendment of Plan

The Board reserves the right at any time and from time to time, and retroactively if deemed necessary or appropriate, to amend in whole or in part any or all of the provisions of the Plan. However, no amendment shall make it possible for any part of the funds of the Plan to be used for, or diverted to, purposes other than for the exclusive benefit of persons entitled to benefits under the Plan. No amendment shall be made which has the effect of decreasing the balance of the Accounts of any Member or of reducing the nonforfeitable percentage of the balance of the Accounts of a Member below the nonforfeitable percentage computed under the Plan as in effect on the date on which the amendment is adopted or, if later, the date on which the amendment becomes effective.

#### 13.02 Merger or Consolidation

The Board or its delegate may, in its sole discretion, merge this Plan with another qualified plan or transfer a portion of the Plan's assets or liabilities to another qualified plan, subject to any applicable legal requirements. The Plan may not be merged or consolidated with, and its assets or liabilities may not be transferred to, any other plan unless each person entitled to benefits under the Plan would, if the resulting plan were then terminated, receive a benefit immediately after the merger, consolidation, or transfer which is equal to or greater than the benefit he would have been entitled to receive immediately before the merger, consolidation, or transfer if the Plan had then terminated.

#### 13.03 Termination of Plan

(a) The General Assembly of North Carolina may terminate the Plan or completely discontinue contributions under the Plan for any reason at any time. In case of termination or partial termination of the Plan, or complete discontinuance of Employer contributions to the Plan, the rights of affected Members to their Accounts under the Plan as of the date of the termination or discontinuance shall be nonforfeitable. The total amount in each Member's Accounts shall be distributed to him or for his benefit or continued in trust for his benefit.

(b) Upon termination of the Plan, the value of the Members' Accounts shall be distributed to Members as soon as administratively practicable, provided that (i) the Employer does not establish or maintain a successor defined contribution plan and (ii) payment is made to the Member in the form of a lump sum distribution (as defined in Section 402(e)(4) of the Code, without regard to without regard to subclauses (I) through (IV) of clause (i) thereof). For purposes of this paragraph, a "successor defined contribution plan" is a defined contribution plan (other than an employee stock ownership plan as defined in Section 4975(e)(7) of the Code ("ESOP"), a simplified employee pension as defined in Section 408(k) of the Code ("SEP"), a SIMPLE IRA plan as described in Section 408(p) of the Code, a plan or contract that satisfies the requirements of Section 403(b) of the Code, or a plan that is described in Section 457(b) or (f)) which exists at the time the Plan is terminated or within the 12 month period beginning on the date all assets are distributed that accepts salary deferrals. However, in no event shall a defined contribution plan be deemed a successor plan if fewer than 2 percent of the employees who are eligible to participate in the Plan at the time of its termination are or were eligible to participate under another defined contribution plan of the Employer or an Affiliated Employer (other than a plan excluded under the prior sentence) at any time during the period beginning 12 months before and ending 12 months after the date of the Plan's termination.

#### 13.04 Merger of Jackson County 401(k) Retirement Plan with the Plan.

Effective October 28, 2004, the merger of the Jackson 401(k) Retirement Plan ("Jackson County Plan") into the Plan is approved contingent upon each person entitled to benefits under the Jackson County 401(k) Retirement Plan would, if the resulting plan were then terminated, receive a benefit immediately after the merger, consolidation, or transfer if the Jackson County Plan had then terminated. The merger is further contingent upon the Jackson County Plan complying with all applicable Internal Revenue Code provision relating to the merger. Upon completion of the merger, the terms of the Plan shall control except that any outstanding loans made pursuant to the Jackson County Plan that do not conform with Article 4 of the Plan shall be repaid according to the respective repayment terms of the loan(s).

### 13.05 Merger of Public Works Commission of the City of Fayetteville, North Carolina Defined Contribution Plan.

Effective of October 28, 2004, the merger of the Public Works Commission of the City Fayetteville, North Carolina Defined Contribution Plan ("Public Works Plan") into the Plan is approved contingent upon each person entitled to benefits under the Public Works Plan would, if the resulting plan were then terminated, receive a benefit immediately after the merger, consolidation, or transfer which is equal to or greater than the benefit he or she would have been entitled to receive immediately before the merger, consolidation, or transfer if the Public Works Plan had then terminated. The merger is further contingent upon the Public Works Plan complying with all applicable Internal Revenue Code provisions relating to the merger. Upon completion of the merger, the terms of the Plan shall control except that any outstanding loans made pursuant to the Public Works Plan that do not conform with Article 4 of this Plan shall be repaid according to the respective repayment terms of the loan(s).

### 13.06 Merger of the Town of Spring Lake, North Carolina Deferred Salary Profit Sharing Plan with the Plan.

Effective April 21, 2005, the merger of The Town of Spring Lake, North Carolina Deferred Salary Profit Sharing Plan ("Spring Lake Plan") into the Plan is approved contingent upon each person entitled to benefits under the Spring Lake Plan would, if the resulting plan were then terminated, receive a benefit immediately after the merger, consolidation, or transfer which is equal to or greater than the benefit he or she would have been entitled to receive immediately before the merger, consolidation, or transfer if the Spring Lake Plan had then terminated. The merger is further contingent upon the Spring Lake Plan complying with all applicable Internal Revenue Code provision relating to the merger. Upon completion of the merger, the terms of the Plan shall control.

#### APPENDIX A

- 1. Subject to the provisions of Section 2.01, an Employee, not including State employed law enforcement officers, shall be eligible to become a Member of Part "A" on the first Enrollment Date on or after the date he becomes
  - (i) a member of the
    - (1) Teachers' and State Employees' Retirement System of North Carolina;
    - (2) the Consolidated Judicial Retirement System; or
    - (3) the Legislative Retirement System, or
  - (ii) a participant in the Optional Retirement Program.

The eligibility requirements for membership in the above Retirement Systems are as follows:

- (i) As provided in General Statutes of North Carolina 135-3, employees and teachers employed by local boards of education, employees and teachers employed by the State's university system and State employees, shall become members in the Teachers' and State Employees' Retirement System immediately upon their election, appointment or employment. On January 1, 1985, the members of the Law Enforcement Officers' Retirement System who were State law enforcement officers became members of the Teachers' and State Employees' Retirement System.
- (ii) As provided in General Statutes of North Carolina 135-55, all judges, district attorneys in office, or clerks of the Superior Court shall be members of the Judicial Retirement System.
- (iii) As provided in General Statutes of North Carolina 120-4.11, all members of the General Assembly who are not active members of any of the following retirement systems shall be members of the Legislative Retirement System:
  - (a) The Teachers' and State Employees' Retirement System;
  - (b) The North Carolina Local Governmental Employees' Retirement System; and
  - (c) The Consolidated Judicial Retirement System.
- (iv) As provided in the General Statutes of North Carolina 135-5.1, participants in the Optional Retirement Program immediately prior to July 1, 1985 shall continue to be participants and all other administrators and faculty of the University of North Carolina with rank of instructor or above shall become participants upon their election at the time they enter eligible employment.

- 2. Subject to the provisions of Section 2.01, an Employee shall be eligible to become a Member of Part "B" on the first Enrollment Date on or after the date he becomes
  - (i) a member of the Local Governmental Employees' Retirement System;
  - (ii) employed as a law enforcement officer by any political subdivision of the State; or
  - (iii) a member of a retirement or pension plan qualified under Section 401(a) of the Code which is sponsored by a political subdivision of the State of North Carolina.

The eligibility requirements for such membership are as follows:

As provided in General Statutes of North Carolina 128-24, all employees entering or reentering the service of a participating county, city or town after the date of participation in the retirement system of such county, city or town, may become members of the North Carolina Local Governmental Employees' Retirement System. On January 1, 1986 the members of the Law Enforcement Officers' Retirement System who were local law enforcement officers became members of the Local Governmental Employees' Retirement System.

3. Subject to the provisions of Section 2.01, an Employee shall be eligible to become a Member of Part "C" on the first Enrollment Date on or after the date he becomes employed by the State of North Carolina as a law enforcement officer.

## PRUDENTIAL RETIREMENT SPECIMEN GOVERNMENTAL 457(b) PLAN ADOPTION AGREEMENT

This Adoption Agreement is to be used in conjunction with the Prudential Retirement

Specimen Governmental 457(b) Plan Document

This Adoption Agreement is <u>not</u> to be used for a tax exempt organization 457(b) plan.

This document has not been amended to reflect specific provisions of applicable state law. Each adopting entity should review its local and state law requirements before either adopting the Prudential Retirement Specimen Governmental 457(b)

Plan or executing this Adoption Agreement.

Prudential and its affiliates do not provide tax or legal advice. This specimen document is for the use of the Employer's tax or legal advisors.

#### ADOPTION AGREEMENT

#### Part I Adopting Entity

Name of Employer: State of North Carolina

Address: 3200 Atlantic Ave

**Supplemental Retirement Plans, Second Floor** 

City: Raleigh

State: North Carolina

Zip: **27604** 

Employer Contact: **Steve Toole** 

**Contact Phone** 

Number: **919-814-4197** 

Employer's Tax ID: **60-0557114** 

#### **ADOPTION AGREEMENT**

#### Part II Plan Administrator

| Name of Plan                      |                                             |
|-----------------------------------|---------------------------------------------|
| Administrator:                    | State of North Carolina                     |
|                                   |                                             |
| Address:                          | 3200 Atlantic Ave                           |
|                                   | Supplemental Retirement Plans, Second Floor |
| City:                             | Raleigh                                     |
| State:                            | North Carolina                              |
| Zip:                              | 27604                                       |
| Plan Administrator                |                                             |
| Phone Number:                     | 919-814-4197                                |
| Plan Administrator<br>Fax Number: |                                             |
| Plan Administrator                |                                             |
|                                   | Steve Toole                                 |
| E-mail (if                        |                                             |
| available):                       | steve.toole@nctreasurer.com                 |

#### ADOPTION AGREEMENT

|       |                    | Part III Plan Data                                                                                                                                                                                                                                  |   |  |
|-------|--------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---|--|
| 1.    | This               | Plan is (check one):   A new plan.  X A restatement of an existing plan.                                                                                                                                                                            |   |  |
| Cauı  | tion:              | If this is a restatement of an existing plan, the Employer should carefully review the prior Plan document to ensure that any changes from the terms of the prior Plan document which may result from the adoption of the restatement are intended. |   |  |
| 2.    | Nam                | ne of Plan. The name of the Plan is (provide name):                                                                                                                                                                                                 |   |  |
|       |                    | North Carolina Public Employee Deferred CompensationPlan                                                                                                                                                                                            | _ |  |
| 3.    | <u>Plan Year</u> . |                                                                                                                                                                                                                                                     |   |  |
|       | (a)                | The Plan Year is (a twelve month period – <u>e.g.</u> , Jan. 1.– Dec. 31):                                                                                                                                                                          |   |  |
|       |                    | January 1 – December 31                                                                                                                                                                                                                             |   |  |
|       | (b)                | The Plan $\square$ will $/\square$ will not ( <i>check one</i> ) have a short initial Plan Year. The short Plan Year will be ( <i>a less than twelve month period</i> $-$ <i>e.g., Oct.</i> $1 -$ <i>Dec.</i> $31$ ):                               |   |  |
|       |                    | N/A                                                                                                                                                                                                                                                 |   |  |
| 4.    |                    | ctive Date. vide information required by Item $4(a)$ or $4(b)$ (as applicable)):                                                                                                                                                                    |   |  |
|       | (a)                | For new plans:                                                                                                                                                                                                                                      |   |  |
|       |                    | (i) The Effective Date of the Plan is ( <i>provide date</i> ):                                                                                                                                                                                      |   |  |
| (b) I |                    | For restated plans:                                                                                                                                                                                                                                 |   |  |
|       |                    | (i) The initial effective date of the Plan was ( <i>provide date</i> ):                                                                                                                                                                             |   |  |

#### **ADOPTION AGREEMENT**

# November 12, 1974 (ii) The Effective Date of this amendment and restatement of the Plan is (provide date): December 10, 2015

#### **ADOPTION AGREEMENT**

#### **Part IV** Eligibility Requirements

| 5. | Eligible Employees. |                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
|----|---------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|    | (a)                 | The following Employees of the Employer are eligible to participate in the Plan ( <i>check one</i> ):                                                                                                                                                                                                                                                                                                         |  |  |
|    |                     | <ul> <li>□ All employees.</li> <li>X Only the following classes of employees (<i>please specify</i>):</li> </ul>                                                                                                                                                                                                                                                                                              |  |  |
|    |                     | Any employee of an Adopting Employer to include employees of the State of North Carolina, any county or municipality of the State, the North Carolina Community College System, and any political subdivision of the State. This shall include, but not be limited to, full-time and part-time employees, elected and appointed officials and re-employed retired employees, except as excluded in (b) below. |  |  |
|    |                     | ☐ All employees other than the following classes ( <i>please specify</i> ):                                                                                                                                                                                                                                                                                                                                   |  |  |
|    |                     | Note: The definition for an Adopting Employer is considered as the following: i) State of North Carolina (the "State") and its agencies, departments, subdivisions and instrumentalities; ii) counties and municipalities of the State and any other political subdivisions of the State; and iii) Community Colleges of the State and any local board of education which is electing to adopt the Plan.      |  |  |
|    | (b)                 | Independent contractors performing services for the Employer who satisfy the eligibility criteria selected in Item $5(a) \square$ are / $X$ are not ( <i>check one</i> ) permitted to participate in the Plan.                                                                                                                                                                                                |  |  |
| 6. | Entry               | <u>Date</u> . The following is the Plan's Entry Date ( <i>check one</i> ):                                                                                                                                                                                                                                                                                                                                    |  |  |
|    | X                   | The first day of the Plan Year and the first day of each calendar month thereafter.                                                                                                                                                                                                                                                                                                                           |  |  |
|    |                     | The first day of the Plan Year and the first day of each calendar quarter thereafter.                                                                                                                                                                                                                                                                                                                         |  |  |
|    |                     | The first day of the Plan Year and the first day of the seventh (7th) month of the Plan Year.                                                                                                                                                                                                                                                                                                                 |  |  |
|    |                     | Other (please specify):                                                                                                                                                                                                                                                                                                                                                                                       |  |  |
|    |                     |                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |

#### ADOPTION AGREEMENT

#### **Part V** Elective Deferrals

| 7 | • | Includible | Com | pensation. |
|---|---|------------|-----|------------|
|   |   |            |     |            |

- (a) Except as provided below in the case of independent contractors, Includible Compensation means (*check one*):
  - □ Safe Harbor Compensation as described in Code section 415(c)(3) and the regulations promulgated thereunder.
  - Code section 3401(a) wages (as modified by Code section 415(c)(3) and the regulations promulgated thereunder).
  - □ Information required to be reported under Code sections 6041, 6051, and 6052 (e.g., the "Wages, Tips and Other Compensation" box on Form W-2) (as modified by Code section 415(c)(3) and the regulations promulgated thereunder).
  - **X** Other (please specify a definition that complies with the requirements of Code section 415(c)(3) and the regulations promulgated thereunder (including any optional inclusions/exclusions under Code section 415(c)(3)):

Information required to be reported under Code sections 6041, 6051, and 6052 (e.g., "Wages, Tips and other Compensation" bos on Form W-2 (as modified by Code section 415(c)(3)) and to include post-severance payments as defined by the final section 415 rules and as may be deferred and treated as deemed compensation under Section 415.

The Includible Compensation of independent contractors eligible to participate in the Plan pursuant to an election under Item 5(b) shall be determined in accordance with the special rules for self-employed individuals in Code section 415(c)(3)(B) and the regulations promulgated thereunder.

(b) Includible Compensation □ does / **X** does not (*check one*) include amounts treated as "deemed 125 compensation" because of the Employer's requirement that its Employees participate in an Employer-sponsored health insurance program unless they state that they are provided health care coverage elsewhere.

#### ADOPTION AGREEMENT

Note: Pursuant to the Heroes Earnings Assistance and Relief Tax Act of 2008, amounts paid as differential military pay are included in the Internal Revenue Code definition of includible compensation.

| (a) The minimum amount and/or percentage by which a Participan to have his or her Includible Compensation reduced and contrib Plan as Deferred Compensation is equal to ( <i>check one</i> ): |               |                                                                        |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------|------------------------------------------------------------------------|
|                                                                                                                                                                                               | X             | There is no minimum amount or percentage.                              |
|                                                                                                                                                                                               |               | Other (complete Items $8(a)(i)$ and/or $8(a)(ii)$ , as necessary):     |
|                                                                                                                                                                                               |               | (i) \$                                                                 |
|                                                                                                                                                                                               |               | (ii)% of Includible Compensation                                       |
| (b)                                                                                                                                                                                           | The 1         | limits described in Item 8(a) shall be applied on the following basis: |
|                                                                                                                                                                                               | □<br><b>X</b> | Annually. Monthly. Pay Period by Pay Period Other (please specify)     |

### 9. Maximum Deferrals.

8.

- (a) The maximum annual amount and/or percentage by which a Participant may elect to have his or her Includible Compensation reduced and contributed to the Plan as Deferred Compensation (other than as a catch-up contribution) is equal to (*check one*):
  - X The Dollar Limitation (<u>e.g.</u>, the amounts set forth in the chart below) and Percentage Limitation (<u>e.g.</u>, 100% of Includible Compensation) in effect under Code section 457(b)(2) for the applicable Plan Year.

#### ADOPTION AGREEMENT

**Dollar Limitation** 

Year

10.

|               |        | <u>1 Ear</u>                                             |              | ©15.500                                                                                                 |
|---------------|--------|----------------------------------------------------------|--------------|---------------------------------------------------------------------------------------------------------|
|               |        | 2008                                                     |              | \$15,500                                                                                                |
|               |        | 2009                                                     |              | \$16,500                                                                                                |
|               |        | 2010                                                     |              | \$16,500                                                                                                |
|               |        | 2011                                                     |              | \$16,500                                                                                                |
|               |        | 2012 an                                                  | d later      | As adjusted for the cost-of-living                                                                      |
|               |        | Other (complex                                           | te Items 9(a | a)(i) and 9(a)(ii), as necessary):                                                                      |
|               |        | (i) \$                                                   |              |                                                                                                         |
|               |        | ,                                                        | If no percer | Includible Compensation (not in excess of stage is selected a default maximum will automatically apply. |
| (b)           | The li | mits described i                                         | n Item 9(a)  | shall be applied on the following basis:                                                                |
|               | X      | Annually.                                                |              |                                                                                                         |
|               |        | Monthly.                                                 |              |                                                                                                         |
|               |        | Pay Period by                                            | Pay Period   |                                                                                                         |
|               |        | Other (please s                                          | •            |                                                                                                         |
|               |        |                                                          |              |                                                                                                         |
|               |        |                                                          |              |                                                                                                         |
|               |        |                                                          |              | bes / □ does not ( <i>check one</i> ) permit Roth ontributions will be permitted ( <i>check one</i> ):  |
| □<br><b>X</b> | On an  | nd after January<br>nd after <u>A</u><br>January 1, 2011 | April 1, 201 | (Indicate a date on or                                                                                  |
| Note          |        |                                                          |              | ons will be treated as pre-tax elective                                                                 |

deferral contributions for all purposes under the Plan, including, but not limited to, eligibility for matching contributions. In addition, Roth elective deferral contributions must be permitted under the Plan to permit Roth rollover contributions.

#### ADOPTION AGREEMENT

- 11. <u>Last Three Years Catch-Up Contributions</u>. The maximum annual amount of catch-up contributions available in the last three taxable years before Normal Retirement Age which may be made by any Participant shall not exceed (*check one*):
  - X The Catch-Up Contribution limit (<u>e.g.</u>, double the amounts set forth in the chart in Item 9(a)) in effect under Code section 457(b)(2) for the applicable Plan Year.

| Other (please specify a lower dollar amount): |
|-----------------------------------------------|
|                                               |
|                                               |
|                                               |

12. Age 50 Catch-Up Contributions. The Plan  $\mathbf{X}$  will /  $\square$  will not (*check one*) permit Participants to make Code section 414(v) catch-up contributions up to the amounts set forth in the chart below (as permitted under the rules coordinating last three years catch-up contributions and Code section 414(v) catch-up contributions).

| <u>Year</u>    | Age 50 Catch-up Limit              |
|----------------|------------------------------------|
| 2008           | \$5,000                            |
| 2009           | \$5,500                            |
| 2010           | \$5,500                            |
| 2011           | \$5,500                            |
| 2012 and later | As adjusted for the cost-of-living |

Note: If the Plan permits Roth elective deferral contribution, Roth catch-up contributions will also be permitted and treated as pre-tax catch-up contributions for all purposes under the Plan.

- 13. Accumulated Sick, Vacation, and Back Pay.
  - (a) The Plan  $\mathbf{X}$  does /  $\square$  does not (*check one*) permit Participants to contribute the value of their accumulated sick pay to the Plan.
  - (b) The Plan  $\mathbf{X}$  does /  $\square$  does not (*check one*) permit Participants to contribute the value of their accumulated vacation pay to the Plan.
  - (c) The Plan  $\mathbf{X}$  does /  $\square$  does not (*check one*) permit Participants to contribute the value of their accumulated back pay to the Plan.

### **ADOPTION AGREEMENT**

Note: Such elections may only be made in accordance with the requirements of Code section 457(b) and the regulations promulgated thereunder.

# 14. Automatic Enrollment.

| (a) | The Plan $\square$ does / <b>X</b> does not ( <i>check one</i> ) automatically enroll newly hired Eligible Employees and $\square$ does / <b>X</b> does not ( <i>check one</i> ) automatically enroll rehired Eligible Employees. If automatic enrollment applies under this Item 14(a), Eligible Employees hired or rehired, as applicable, on or after ( <i>indicate date</i> ) will be automatically enrolled thirty (30) days after his or her date of hire at a pre-tax elective deferral contribution rate of % ( <i>indicate default percentage</i> ) of Includible Compensation. Automatic enrollment will take effect as soon as administratively practicable. |
|-----|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| (b) | The Plan $\Box$ does / <b>X</b> does not ( <i>check one</i> ) automatically enroll all Eligible Employees. If automatic enrollment applies under this Item 14(b), Eligible Employees hired on or after                                                                                                                                                                                                                                                                                                                                                                                                                                                                  |
| (c) | The Plan $\Box$ does / <b>X</b> does not ( <i>check one</i> ) permit, to the extent permitted by Code section 414(w), automatically enrolled participants to withdraw contributions made pursuant to an automatic enrollment provision elected under Item 14(a) or 14(b) above if the election is made no later than the date which is 90 days after the date of the Eligible Employee's first elective contribution.                                                                                                                                                                                                                                                   |
|     | Note: These automatic enrollment-related provisions will be interpreted and conformed to comply with any applicable Internal Revenue                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |

Service requirements.

#### ADOPTION AGREEMENT

#### Part VI Loans

### 15. Loans.

- (a) The Plan **X** does / □ does not (*check one*) permit Participants to take loans. If Plan loans are permitted, a copy of the Plan Administrator's loan procedures shall be provided to Prudential.
- (b) The following individuals are eligible to receive Plan loans (*check all that apply*):
  - All Plan Participants (including former Employees with balances in the Plan and Employees not currently eligible to contribute to the Plan).
  - X Only Plan Participants who are eligible to contribute to the Plan because they are eligible employees described in Item 5 at the time the loan is requested.

Revised: July 2011

## **ADOPTION AGREEMENT**

# Part VII Additional Employer Contributions

16.

| Disc | retionary                                                                                                                            | Contributions.                                                                                                                                                                                                                                                                                                                                                                                                |  |  |
|------|--------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
| (a)  | The Employer $X$ may $/ \square$ may not ( <i>check one</i> ) make additional discretionary contributions to Participants' Accounts. |                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
| (b)  |                                                                                                                                      | Participants shall vest in discretionary contributions on the following basis ( <i>check one</i> ):                                                                                                                                                                                                                                                                                                           |  |  |
|      | <b>X</b>                                                                                                                             | Immediate, 100% vesting Graded vesting at a rate of 20% per year (100% at 5 years) Other (please specify an alternative vesting schedule):                                                                                                                                                                                                                                                                    |  |  |
|      | Note:                                                                                                                                | Under regulations issued by the Internal Revenue Service, contributions, and earnings on these contributions that are subject to a vesting schedule are recognized as contributions to the Plan when these amounts vest. As a result, if an Employer elects to apply a vesting schedule, Participants may be deemed to have impermissible excess contributions to the Plan in the year in which contributions |  |  |
|      |                                                                                                                                      | vest if the amounts vested during the year causes an excess deferral for that year. Employers should discuss this issue with independent legal counsel prior to electing to apply a vesting schedule.                                                                                                                                                                                                         |  |  |
|      |                                                                                                                                      | butions for a Participant who dies while in qualified military service ant to and in a manner consistent with Code section 414(u)(9). This                                                                                                                                                                                                                                                                    |  |  |
|      | $\overset{\square}{X}$                                                                                                               | January 1, 2007<br>Other (please specify a date no earlier than January 1, 2007):                                                                                                                                                                                                                                                                                                                             |  |  |
|      |                                                                                                                                      | January 1, 2010                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
| (d)  |                                                                                                                                      | mployer <b>X</b> may / □ will not ( <i>check one</i> ) make discretionary butions for a Participant who becomes disabled while in qualified                                                                                                                                                                                                                                                                   |  |  |

#### ADOPTION AGREEMENT

military service pursuant to and in a manner consistent with Code section 414(u)(9). This provision is effective (*check one*):

- □ January 1, 2007
  X Other (please specify a date no earlier than January 1, 2007):

  January 1, 2010
- 17. Matching Contributions.
  - (a) The Employer **X** may / □ will not (*check one*) match Participant contributions to the Plan. Matching contributions will be made based on the following formula (*please specify*):

As enacted by the North Carolina General Assembly or approved by the Governing Board of a political subdivision of the State.

- (b) Participants shall vest in matching contributions on the following basis (*check one*):
  - X Immediate, 100% vesting
  - □ Graded vesting at a rate of 20% per year (100% at 5 years)
  - $\Box$  Other (please specify an alternative vesting schedule):

Note: Under regulations issued by the Internal Revenue Service, contributions, and earnings on these contributions that are subject to a vesting schedule are recognized as contributions to the Plan when these amounts vest. As a result, if an Employer elects to apply a vesting schedule, Participants may be deemed to have impermissible excess contributions to the Plan in the year in which contributions vest if the amounts vested during the year causes an excess deferral for that year. Employers should discuss this issue with independent legal counsel prior to electing to apply a vesting schedule.

## **ADOPTION AGREEMENT**

| (c) | for a Participant who dies while in qualified military service pursuant to and in a manner consistent with Code section $414(u)(9)$ . This provision is effective ( <i>check one</i> ): |                                                                                                                                                                                                                                                                                   |  |  |
|-----|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|     |                                                                                                                                                                                         | January 1, 2007                                                                                                                                                                                                                                                                   |  |  |
|     | X                                                                                                                                                                                       | Other (please specify a date no earlier than January 1, 2007):                                                                                                                                                                                                                    |  |  |
|     |                                                                                                                                                                                         | January 1, 2010                                                                                                                                                                                                                                                                   |  |  |
| (d) | for a                                                                                                                                                                                   | Employer <b>X</b> may $/ \square$ will not ( <i>check one</i> ) make matching contributions a Participant who becomes disabled while in qualified military service uant to and in a manner consistent with Code section 414(u)(9). This rision is effective ( <i>check one</i> ): |  |  |
|     |                                                                                                                                                                                         | January 1, 2007                                                                                                                                                                                                                                                                   |  |  |
|     | X                                                                                                                                                                                       | Other (please specify a date no earlier than January 1, 2007):                                                                                                                                                                                                                    |  |  |
|     |                                                                                                                                                                                         | January 1, 2010                                                                                                                                                                                                                                                                   |  |  |

Revised: July 2011

## **ADOPTION AGREEMENT**

# Part VIII Retirement Age and Commencement of Distributions

| 18. | the La | Normal Retirement Age. A Participant's Normal Retirement Age for purposes of the Last Three Years Catch-Up Contributions described in Item 11 of this adoption agreement is ( <i>check one</i> ):                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |  |
|-----|--------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|     |        | Age 70½.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                          |  |  |  |
|     |        | Age 65                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |
|     |        | Age 62                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |
|     |        | Age 60                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |
|     |        | Age 55                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |
|     | X      | Other (please specify):                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
|     |        | Normal retirement age is defined as the date that the member would be eligible to retire with an unreduced service retirement from the State Administered Retirement System in which the member currently participates, but not later than age 70 ½.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                              |  |  |  |
|     | Note:  | A Participant's Normal Retirement Age may not be (a) earlier than the earlier of age 65 or the earliest date that the Participant will become eligible to retire and receive under the Employer's basic defined benefit pension plan covering the Participant (or the Employer's money purchase pension plan in which the Participant also participates if the Participant is not eligible to participate in the Employer's basic defined benefit pension plan) immediate retirement benefits without actuarial or similar reduction because of retirement before some later specified age, if any, and (b) later than the Participant's mandatory retirement age (or age 70-1/2, if earlier), if any. Special rules may apply in the case of "qualified" police or firefighters. |  |  |  |
| 19. |        | natic Distribution Date. The date on which the distribution of a Participant's enefit must commence is ( <i>check one</i> ):                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
|     |        | The latest date permitted under Code section 401(a)(9)                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |
|     |        | 60 days after the close of the Plan Year in which a Participant's Severance                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |  |  |
|     | _      | Event occurs                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                      |  |  |  |
|     |        | Age 70½                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |  |  |  |
|     |        | Age 65                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |
|     |        | Age 62                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |
|     |        | Age 60                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |  |

#### ADOPTION AGREEMENT

Note:

20.

| $\mathbf{X}$             | Age<br>Othe                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    | 55<br>r (please specify):                                                                                                                                                                                                                                                                                               |
|--------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| North                    | erance<br>Caro                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 | Participants with account balances of less than \$1000 and who have had Event, distributions shall be made upon the direction of the State of lina. For all other Participants, benefits must commence on the latest ted under Code Section 401(a)(9).                                                                  |
| will n<br>Sever<br>may r | ot apprance In the second seco | of the election made in this Item 19, the Automatic Distribution Date by to a Participant until he or she either (1) attains age 70½ or (2) has a Event. In addition, the date selected as an Automatic Distribution Date later than the latest date permitted under Code section 401(a)(9) and promulgated thereunder. |
| <u>Other</u>             | Distri                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                         | ibutions Events.                                                                                                                                                                                                                                                                                                        |
| (a)                      | distri<br>conti                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | Plan <b>X</b> does / $\square$ does not ( <i>check one</i> ) permit a Participant to receive ibutions of pre-tax contributions (including pre-tax rollover ributions) in-service starting in the calendar year in which the cipant attains age $70\frac{1}{2}$ .                                                        |
| (b)                      | The Plan <b>X</b> does / $\square$ does not ( <i>check one</i> ) permit a Participant to receive distributions of Roth contributions (including Roth rollover contributions) in-service starting in the calendar year in which the Participant attains age $70\frac{1}{2}$ .                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                         |
| (c)                      | The Plan \( \precedot\) does / <b>X</b> does not ( <i>check one</i> ) permit a Participant to receive distributions while in military service pursuant to and in a manner consistent with the Heroes Earnings Assistance and Relief Tax Act of This provision is effective ( <i>check one</i> ):                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |                                                                                                                                                                                                                                                                                                                         |
|                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                | January 1, 2009 Other (please specify a date no earlier than January 1, 2009):                                                                                                                                                                                                                                          |

Any distributions pursuant to this Item 20(c) shall result in a suspension of a Participant's right to contribute Deferred Compensation for a period of six-months from the date of distribution in a manner consistent with Code section 414(u)(12)(B).

#### ADOPTION AGREEMENT

### **Part IX Distribution Options**

### 21. Normal Distribution.

(a) A Participant's Account may be paid in any of the following payment forms (*check all that apply*):

**X** Option 1: One lump-sum payment;

X Option 2: Equal monthly, quarterly, semi-annual or annual

payments in an amount chosen by the Participant, continuing until his or her Account is exhausted;

**X** Option 3: Approximately equal monthly, quarterly, semi-annual

or annual payments, calculated to continue for a period

certain chosen by the Participant.

□ Option 4: Annual Payments equal to the minimum distributions

required under Code section 401(a)(9), including the incidental death benefit requirements of Code section

401(a)(9)(G), over the life expectancy of the Participant or over the life expectancies of the

Participant and his or her Beneficiary.

X Option 5: Payments equal to payments made by the issuer of a

retirement annuity policy acquired by the Employer provided that such payments shall comply with the requirements of Code section 401(a)(9), including the incidental death benefit requirements of Code section

401(a)(9)(G).

**X** Option 6 Other (*please specify*):

### Partial lump sum (\$500 minimum)

**X** Option 7: A split distribution under which payments under

Options 1, 2, 3, 5, or 6 commence or are made at the

same time.

**X** Option 8: A split distribution under which payments under

Options 1, 2, 3, 5, or 6 commence or are made at

different times.

#### ADOPTION AGREEMENT

Note: A Participant may not elect a payment form which causes his or her distributions to violate the minimum required distribution requirements of Code section 401(a)(9).

(b) The default method of paying benefits to a Participant who has reached his or her Automatic Distribution Date but failed to timely elect to receive a distribution is the distribution method available under Option <u>2 – annual</u> <u>payment</u> (provide Option number) of Item 21(a).

Note: If the default method of paying benefits (1) results in the payment of an "eligible rollover distribution" (*i.e.*, eligible to be rolled over to a 401(a), 403(a), 403(b), governmental 457(b) plan, or an IRA) that exceeds \$1,000 (including rollover contributions), (2) occurs because of a Participant's reaching his or her Automatic Distribution Date prior to the Participant's Normal Retirement Date (or age 62, if later), and (3) occurs on or after the effective date listed in Item 26, the distribution will be subject to the Code section 401(a)(31)(B) mandatory rollover rules.

- 22. <u>Post-Retirement Death Benefits</u>. If a Participant dies after he or she has begun receiving benefits from the Plan, the Participant's Account shall be paid to the Participant's Beneficiary (*check one*):
  - ☐ In accordance with the minimum required distribution rules under Code section 401(a)(9).
  - **X** In a single lump sum as soon as administratively practicable.

#### 23. Pre-Retirement Death Benefits.

(a) If a Participant dies before he or she has begun receiving benefits from the Plan, the Participant's Beneficiary may elect to have the Participant's Account paid in one of the following payment forms (*check all that apply*):

**X** Option 1: One lump-sum payment;

**X** Option 2: Equal monthly, quarterly, semi-annual or annual

payments in an amount chosen by the Beneficiary, continuing until his or her Account is exhausted;

X Option 3: Approximately equal monthly, quarterly, semi-annual

or annual payments, calculated to continue for a period

certain chosen by the Beneficiary.

## **ADOPTION AGREEMENT**

|                                                                                                                                                                           | Option 4:                                                                                                                                                               | Annual Payments equal to the minimum distributions required under Code section 401(a)(9), including the incidental death benefit requirements of Code section 401(a)(9)(G), over the life expectancy of the                                                                      |  |
|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|
| X                                                                                                                                                                         | Option 5:                                                                                                                                                               | Beneficiary. Payments equal to payments made by the issuer of a retirement annuity policy acquired by the Employer provided that such payments shall comply with the requirements of Code section 401(a)(9), including the incidental death benefit requirements of Code section |  |
| X                                                                                                                                                                         | Option 6                                                                                                                                                                | 401(a)(9)(G). Other (please specify):                                                                                                                                                                                                                                            |  |
|                                                                                                                                                                           | Partia                                                                                                                                                                  | d lump sum (\$500 minimum)                                                                                                                                                                                                                                                       |  |
| X                                                                                                                                                                         | Option 7:                                                                                                                                                               | A split distribution under which payments under Options 1, 2, 3, 5, or 6 commence or are made at the same time.                                                                                                                                                                  |  |
| X                                                                                                                                                                         | Option 8:                                                                                                                                                               | A split distribution under which payments under Options 1, 2, 3, 5, or 6 commence or are made at different times.                                                                                                                                                                |  |
| Note:                                                                                                                                                                     | : A Beneficiary may not elect a payment form which causes his or her distributions to violate the minimum required distribution requirements of Code section 401(a)(9). |                                                                                                                                                                                                                                                                                  |  |
| The default method of paying benefits to a Beneficiary when a Participant has died before payment of his or her Plan benefit has commenced shall be ( <i>check one</i> ): |                                                                                                                                                                         |                                                                                                                                                                                                                                                                                  |  |
| □<br>□<br><b>X</b>                                                                                                                                                        | A lump sum.<br>A single life<br>Other ( <i>please</i>                                                                                                                   | · · · · · · · · · · · · · · · · · · ·                                                                                                                                                                                                                                            |  |
|                                                                                                                                                                           | Annu                                                                                                                                                                    | al installments as available under Option 2, above                                                                                                                                                                                                                               |  |

(c) Distribution of the benefits described in this Item 23 shall commence (*check one*):

(b)

## ADOPTION AGREEMENT

24.

25.

|       | <b>X</b>                                                                                                                                                                                                                                                                                                                                                                                                  | On the date selected by the Beneficiary so long as such date complies with the distribution requirements of Code section 401(a)(9). If the Beneficiary fails to make an election before the (please provide the number of days, default is 60th day) day following the death of the Participant, the distribution of benefits described in this Item 23 shall commence on the (please provide the number of days, default is the 61st day) day following the Participant's death.  On the (please provide the number of days, default is the 61st day) day following the Participant's death. |  |  |
|-------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
| Unfor | reseeabl                                                                                                                                                                                                                                                                                                                                                                                                  | e Emergencies.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                |  |  |
| (a)   | The Plan $X$ does / $\square$ does not ( <i>check one</i> ) permit Participants to receive distributions in the event of Unforeseeable Emergencies. Unforeseeable Emergency distributions may be made from pre-tax contributions (including pre-tax rollover contributions) and $X$ may / $\square$ may not ( <i>check one</i> ) be made from Roth contributions (including Roth rollover contributions). |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
| (b)   |                                                                                                                                                                                                                                                                                                                                                                                                           | rticipants $\square$ may / <b>X</b> may not ( <i>check one</i> ) receive a withdrawal under m 24(a) based on Beneficiaries' Unforeseeable Emergencies.                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
| (c)   | Partici                                                                                                                                                                                                                                                                                                                                                                                                   | The Plan $\square$ does / <b>X</b> does not ( <i>check one</i> ) permit Beneficiaries of deceased Participants to receive distributions in the event of Unforeseeable Emergencies.                                                                                                                                                                                                                                                                                                                                                                                                            |  |  |
| Volur | ntary In-                                                                                                                                                                                                                                                                                                                                                                                                 | Service Distributions.                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                        |  |  |
| (a)   | The Plan $\mathbf{X}$ does / $\square$ does not ( <i>check one</i> ) permit voluntary in-service distributions when the provisions of applicable Treasury Regulations permitting such distributions are satisfied.                                                                                                                                                                                        |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                               |  |  |
| (b)   | eligibl                                                                                                                                                                                                                                                                                                                                                                                                   | Plan permits voluntary in-service distributions, a Participant will be e to elect to receive a voluntary in-service distribution if the value of rticipant's Account is no greater than ( <i>check one</i> ):                                                                                                                                                                                                                                                                                                                                                                                 |  |  |
|       | □<br>□<br><b>X</b>                                                                                                                                                                                                                                                                                                                                                                                        | \$1,000<br>\$3,500<br>\$5,000                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                 |  |  |

#### ADOPTION AGREEMENT

26.

|       | □ Other (please specify an amount not in excess of \$5,000):                                                                                                                                                                                                                                                                                                                     |
|-------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| (c)   | For purposes of calculating the value of a Participant's Account for purposes of applying Item 25(b), a Participant's Account $\mathbf{X}$ will / $\square$ will not ( <i>check one</i> ) include amounts attributable to rollover contributions (as defined in Code section 411(a)(11)(D)).                                                                                     |
| Invol | untary In-Service Distributions.                                                                                                                                                                                                                                                                                                                                                 |
| (a)   | The Plan $\Box$ does / <b>X</b> does not ( <i>check one</i> ) permit involuntary in-service distributions when the provisions of applicable Treasury Regulations permitting such distributions are satisfied. Such distributions shall be made pursuant to a uniform procedure established by the Plan Administrator that is applied in a uniform and non-discriminatory manner. |
| (b)   | If the Plan permits involuntary in-service distributions, a Participant will receive a mandatory in-service distribution if the value of the Participant's Account is no greater than ( <i>check one</i> ):                                                                                                                                                                      |
|       | □ \$1,000 □ \$3,500 □ \$5,000 □ Other (please specify an amount not in excess of \$5,000):  N/A                                                                                                                                                                                                                                                                                  |
| (c)   | For purposes of calculating the value of a Participant's Account for purposes of applying Item 26(b), a Participant's Account $\mathbf{X}$ will / $\square$ will not ( <i>check one</i> ) includes amounts attributable to rollover contributions (as defined in Code section 411(a)(11)(D)).                                                                                    |
|       | Note: If an involuntary distribution (1) results in the payment of an "eligible rollover distribution" ( <i>i.e.</i> , eligible to be rolled over to a 401(a), 403(a), 403(b), governmental 457(b) plan, or an IRA) that                                                                                                                                                         |

exceeds \$1,000 (including rollover contributions), (2) occurs prior to the Participant's Normal Retirement Date (or age 62, if later), and (3) occurs on or after the effective date listed in Item 27, the distribution

### **ADOPTION AGREEMENT**

will be subject to the Code section 401(a)(31)(B) mandatory rollover rules.

|              | ndatory Rollovers. The Plan is subject to the Code section 401(a)(31)(B) adatory rollover rules as of ( <i>check one</i> ):                                                                                                                                                                                    |  |  |
|--------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|              | March 28, 2005                                                                                                                                                                                                                                                                                                 |  |  |
|              | January 1, 2006                                                                                                                                                                                                                                                                                                |  |  |
| X            | Other (please specify a date no later than the close of the first regular legislative session of the legislative body with the authority to amend the Plan that begins on or after January 1, 2006):                                                                                                           |  |  |
|              | The Plan is not subject to 401(a)(31)(B) since the Plan has not elected the option for involuntary in-service distributions under above Item #26.                                                                                                                                                              |  |  |
| perr<br>inhe | a-Spouse Beneficiary Rollovers. The Plan $\mathbf{X}$ does $/ \square$ does not ( <i>check one</i> ) mit non-spouse Beneficiaries to elect to roll over payments owing to them to an erited IRA to the extent permitted under Code section $402(c)(11)$ . This vision is effective as of ( <i>check one</i> ): |  |  |
|              | January 1, 2007                                                                                                                                                                                                                                                                                                |  |  |
|              | January 1, 2008                                                                                                                                                                                                                                                                                                |  |  |
| X            | Other (please specify a date no earlier than January 1, 2007):                                                                                                                                                                                                                                                 |  |  |
|              | December 19, 2008                                                                                                                                                                                                                                                                                              |  |  |
|              | Note: Pursuant to applicable law, non-spouse beneficiary rollovers are mandatory for Plan Years beginning after December 31, 2009.                                                                                                                                                                             |  |  |
| Pre-         | Tax Payment of \$3,000 in Qualified Health Insurance Premiums for Eligible                                                                                                                                                                                                                                     |  |  |
|              | red Public Safety Officers. The Plan $\Box$ does / <b>X</b> does not ( <i>check one</i> ) permit                                                                                                                                                                                                               |  |  |
|              | icipants who are "eligible retired public safety officers" (as defined in Code                                                                                                                                                                                                                                 |  |  |
|              | ion 402(l)(4)(B)) to pay up to \$3,000 in "qualified health insurance premiums"                                                                                                                                                                                                                                |  |  |
| 1            | defined in Code section 402(1)(4)(D)) for the "eligible retired public safety                                                                                                                                                                                                                                  |  |  |

officers", their spouses, or their dependents. This provision is effective as of

(check one):

#### ADOPTION AGREEMENT

|     |             | January 1, 2007                                                                                                                                                                                                                                                                                                  |  |  |
|-----|-------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|
|     |             | January 1, 2008                                                                                                                                                                                                                                                                                                  |  |  |
|     |             | Other (please specify a date no earlier than January 1, 2007):                                                                                                                                                                                                                                                   |  |  |
|     |             | N/A                                                                                                                                                                                                                                                                                                              |  |  |
| 30. | <u>Spou</u> | isal Consent and Default Beneficiary.                                                                                                                                                                                                                                                                            |  |  |
|     | (a)         | The Plan $\Box$ does / <b>X</b> does not ( <i>check one</i> ) require spousal consent for plan distributions of any kind (including loans) in a form other than a joint and survivor annuity.                                                                                                                    |  |  |
|     | (b)         | The Plan $\square$ does / <b>X</b> does not ( <i>check one</i> ) require spousal consent for designating a Beneficiary other than a spouse.                                                                                                                                                                      |  |  |
|     | (c)         | If no Beneficiary is designated by a Participant, the Participant's default beneficiary will be the Participant's $\square$ surviving spouse (if married or estate if unmarried) / $\mathbf{X}$ estate (if married or unmarried) ( <i>check one</i> ).                                                           |  |  |
|     | (d)         | If spousal consent is required pursuant to Item 30(a) or (b), spousal consent must be ( <i>check all that apply</i> ):                                                                                                                                                                                           |  |  |
|     |             | $\square$ Witnessed by $\square$ notary public $/\square$ Plan Administrator ( <i>check all that apply</i> ).                                                                                                                                                                                                    |  |  |
|     |             | □ Irrevocable                                                                                                                                                                                                                                                                                                    |  |  |
|     |             | If spousal consent applies under Item 30(b), spousal consent is required in order to designate any beneficiary $\square$ other than the spouse / $\square$ other than the spouse who separately or combined with other beneficiaries will receive more than 1/2 of the benefits to be paid ( <i>check one</i> ). |  |  |
|     |             | $\Box$ Other (please specify):                                                                                                                                                                                                                                                                                   |  |  |
|     |             | N/A                                                                                                                                                                                                                                                                                                              |  |  |
|     |             | <del></del>                                                                                                                                                                                                                                                                                                      |  |  |

(e) The determination of whether an individual is a person's spouse or surviving spouse at any applicable time **X** is / □ is not (*check one*) made by applying the provisions of the Defense of Marriage Act of 1996. If the Defense of Marriage Act of 1996 is not applied, spouse or surviving spouse status is determined pursuant to procedures attached to this Adoption

#### ADOPTION AGREEMENT

Agreement; provided, however, if no procedures are attached, the provisions of the Defense of Marriage Act of 1996 will apply.

Note: The determination of whether an individual is a person's spouse or surviving spouse at any applicable time is made under procedures adopted by the Plan Administrator. Employers should review the impact of Federal and state laws on spousal beneficiary designations with independent legal counsel prior to completing this Item 30(e).

### **ADOPTION AGREEMENT**

# **Part X** Rollover Contributions

| (a) | The Plan $X$ will / $\square$ will not ( <i>check one</i> ) accept pre-tax rollover contributions from another eligible plan. If pre-tax rollover contributions are accepted the Plan, the pre-tax rollover contributions from the following types of eligible retirement plan shall be accepted by the Plan ( <i>check one</i> ): |                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |  |  |
|-----|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--|--|--|
|     | <b>X</b>                                                                                                                                                                                                                                                                                                                           | All eligible retirement plans as defined in Code section 402(c)(8)( Only governmental Code section 457(b) plans. Other ( <i>please specify</i> ):                                                                                                                                                                                                                                                                                                                     |  |  |  |
|     |                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |  |  |
| (b) | The Plan $X$ will / $\square$ will not ( <i>check one</i> ) accept Roth rollover contributions from another eligible plan. If Roth rollover contributions are accepted by the Plan, the Roth rollover contributions from the following types of eligible retirement plan shall be accepted by the Plan ( <i>check one</i> ):       |                                                                                                                                                                                                                                                                                                                                                                                                                                                                       |  |  |  |
|     | the I                                                                                                                                                                                                                                                                                                                              | n another eligible plan. If Roth rollover contributions are accepted by<br>Plan, the Roth rollover contributions from the following types of                                                                                                                                                                                                                                                                                                                          |  |  |  |
|     | the I                                                                                                                                                                                                                                                                                                                              | a another eligible plan. If Roth rollover contributions are accepted by Plan, the Roth rollover contributions from the following types of ble retirement plan shall be accepted by the Plan ( <i>check one</i> ):  All eligible retirement plans as defined in Code section 402(c)(8)( that maintain a Roth contribution account pursuant to Code section                                                                                                             |  |  |  |
|     | the I<br>eligi                                                                                                                                                                                                                                                                                                                     | n another eligible plan. If Roth rollover contributions are accepted by Plan, the Roth rollover contributions from the following types of ble retirement plan shall be accepted by the Plan ( <i>check one</i> ):  All eligible retirement plans as defined in Code section 402(c)(8)(                                                                                                                                                                                |  |  |  |
|     | the Heligi                                                                                                                                                                                                                                                                                                                         | Plan, the Roth rollover contributions are accepted be Plan, the Roth rollover contributions from the following types of ble retirement plan shall be accepted by the Plan ( <i>check one</i> ):  All eligible retirement plans as defined in Code section 402(c)(8)(8)(1) that maintain a Roth contribution account pursuant to Code section 402A.  Only governmental Code section 457(b) plans that maintain a Roth                                                  |  |  |  |
|     | the I eligi X                                                                                                                                                                                                                                                                                                                      | Plan, the Roth rollover contributions are accepted by Plan, the Roth rollover contributions from the following types of ble retirement plan shall be accepted by the Plan ( <i>check one</i> ):  All eligible retirement plans as defined in Code section 402(c)(8)(6) that maintain a Roth contribution account pursuant to Code section 402A.  Only governmental Code section 457(b) plans that maintain a Roth contribution account pursuant to Code section 402A. |  |  |  |

Roth elective deferral contributions. Roth rollover contributions

#### ADOPTION AGREEMENT

shall be eligible for distribution at the same time as pre-tax rollover contributions.

(c) The following individuals are eligible to make rollover contributions (*check one*):

□ Option #1: All Plan Participants (including former Employees

with balances in the Plan and Employees not currently eligible to contribute to the Plan) and surviving spouse

Beneficiaries.

X Option #2: All Plan Participants (including former Employees

with balances in the Plan and Employees not currently

eligible to contribute to the Plan).

□ Option #3: Only Plan Participants who are otherwise eligible to

contribute to the Plan because they are eligible

employees described in Item 5 at the time they elect to

make a rollover contribution.

## 32. <u>In-Plan Roth</u> Rollovers.

(a) The Plan **X** does / □ does not (*check one*) permit in-plan Roth rollovers (*i.e.*, the conversion of eligible pre-tax funds held in the Plan into Roth after-tax contributions inside the Plan) to the extent permitted under applicable Internal Revenue Service guidance. This provision is effective as of (*check one*):

□ On and after January 1, 2011

X On and after <u>April 1, 2011</u> (*Indicate a date on or after January 1, 2011*)

Note: Roth elective deferral contributions must be permitted under the Plan to permit in-plan Roth rollovers.

(b) The following individuals are eligible to make in-plan Roth rollovers (*check one*):

□ Option #1: All Plan Participants (including former Employees

with balances in the Plan and Employees not currently eligible to contribute to the Plan) and surviving spouse

#### ADOPTION AGREEMENT

Beneficiaries (check only if Option #1 in Item 31(c) is

checked).

 $\mathbf{X}$ All Plan Participants (including former Employees Option #2:

> with balances in the Plan and Employees not currently eligible to contribute to the Plan) (check only if Option

#1 or #2 in Item 31(c) is checked).

Only Plan Participants who are otherwise eligible to Option #3: 

contribute to the Plan because they are eligible

employees described in Item 5 at the time they elect to make a rollover contribution (check only if Option #1,

#2, or #3 in Item 31(c) is checked).

Revised: July 2011

#### ADOPTION AGREEMENT

### Part XI Custody of Assets

- 33. Exclusive Benefit Requirement of Code Section 457(g). The assets of the Plan shall be held for the exclusive benefit of Participants and their beneficiaries. The assets of the Plan shall be held in (*check one or more*):
  - **X** A trust.
  - **X** Custodial accounts within the meaning of Code section 401(f).
  - **X** Annuity contracts within the meaning of Code section 401(f).
- 34. <u>Trustee</u>. If pursuant to Item 33, some or all assets of the Plan are to be held in a trust, the Trustee of the trust shall be (*check one*):
  - □ Prudential Trust Company 30 Scranton Office Park Scranton, PA 18507-1789
  - Prudential Bank & Trust, FSB
     280 Trumbull Street
     Hartford, CT 06013-3513
  - **X** Other:

### **Supplemental Retirement Board of Trustees**

Unless Prudential Trust Company or Prudential Bank & Trust, FSB is the Trustee, this Adoption Agreement shall not become applicable unless the Trustee or a qualified representative thereof approves and countersigns this Adoption Agreement.

# **ADOPTION AGREEMENT**

|     | Part XII Governing Law |                                                                    |  |  |  |
|-----|------------------------|--------------------------------------------------------------------|--|--|--|
| 35. | Gov                    | erning Law. The Plan shall be governed by the laws of (check one): |  |  |  |
|     |                        | New Jersey                                                         |  |  |  |
|     |                        | New York                                                           |  |  |  |
|     |                        | Pennsylvania                                                       |  |  |  |
|     | X                      | Other (please specify which state's laws shall govern the Plan):   |  |  |  |
|     |                        | North Carolina                                                     |  |  |  |

# **ADOPTION AGREEMENT**

| Signatures                                                                                                                                                                                                                                                                                   |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                  |  |  |  |
|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|--|--|--|
| governmental entity describes the Prudential Retirement Seconditions of such Plan and acknowledges that it has reladoption Agreement, (5) unadvisors of the legal and othe limited to, the requirements revise the Plan and Adoption comply with all applicable stacknowledges that Prudenti | greement the Employer hereby certifies that it (1) bed in Code section 457(e)(1)(A), (2) has received specimen Governmental 457(b) Plan, (3) accepts the any related services agreement between it and Prolied upon its own advisors regarding the completion inderstands or has been advised by its own legal and the entire tax implications of adopting the Plan, including of Code section 457(b), (6) has had legal counsel on Agreement to ensure that the Plan and Adoption state and local law requirements, and (7) understand will be under no obligation to update this Adopal Retirement Specimen Governmental 457(b) Plantin applicable law. | a copy of<br>ne terms and<br>udential, (4)<br>on of this<br>ad tax<br>g, but not<br>review and<br>a Agreement<br>nds and<br>tion |  |  |  |
|                                                                                                                                                                                                                                                                                              | the Employer has caused this Adoption Agreemented representative this day of _                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                            |                                                                                                                                  |  |  |  |
|                                                                                                                                                                                                                                                                                              | State of North Carolina  Name of Political Subdivision                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                    |                                                                                                                                  |  |  |  |
| By:                                                                                                                                                                                                                                                                                          |                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                                           |                                                                                                                                  |  |  |  |

Title:

### **ADOPTION AGREEMENT**

Acknowledged on behalf of Prudential Retirement, a business division of The Prudential Insurance Company of America in its role as recordkeeper for the Plan:

| By:    |   |  |  |  |
|--------|---|--|--|--|
|        |   |  |  |  |
|        |   |  |  |  |
| Title: | - |  |  |  |
|        |   |  |  |  |
| ъ.     |   |  |  |  |
| Date:  |   |  |  |  |

Revised: July 2011

## **ADOPTION AGREEMENT**

| • • •   | Agreement, the Trustee hereby acknowledges his one of the Trust Agreement between the Employer |  |
|---------|------------------------------------------------------------------------------------------------|--|
|         | the Trustee has caused this Adoption Agreement ded representative this day of                  |  |
| Attest: | Supplemental Retirement Board of Trustees Name of Trustee (if a Corporate Trustee)             |  |
| Ву:     |                                                                                                |  |
| Title:  |                                                                                                |  |
| By:     |                                                                                                |  |
| Title:  |                                                                                                |  |
| Ву:     |                                                                                                |  |
| Title:  |                                                                                                |  |
| Ву:     |                                                                                                |  |
| Title:  |                                                                                                |  |